UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): April 9, 2021

VAALCO Energy, Inc.

(Exact name of registrant as specified in its charter)

Commission File No. 001-32167

Delaware (State or other jurisdiction of incorporation)

9800 Richmond Avenue, Suite 700 Houston, Texas (Address of principal executive offices) 76-0274813 (I.R.S. Employer Identification No.)

> 77042 (Zip Code)

(713) 623-0801

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

D Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$0.10	EGY	New York Stock Exchange
Common Stock, par value \$0.10	EGY	London Stock Exchange

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter)

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Resignation of Chief Executive Officer, Chief Operating Officer and Director

On April 9, 2021, Mr. Cary M. Bounds and VAALCO Energy, Inc. (the "Company"), entered into a Separation and Mutual Release Agreement (the "Separation Agreement"). Pursuant to the Separation Agreement, Mr. Bounds will resign from his positions as Chief Executive Officer, Chief Operating Officer and a member of the Company's Board of Directors (the "Board") (and any committees thereof) on April 18, 2021. Mr. Bounds will continue to be employed by the Company to assist with the transition of his duties until 5 p.m. on April 30, 2021 (the "Separation Date"). Pursuant to the Separation Agreement, Mr. Bounds will continued to the Separation Agreement, Mr. Bounds will continue to be employed by the Company to assist with the transition of his duties until 5 p.m. on April 30, 2021 (the "Separation Date"). Pursuant to the Separation Agreement, Mr. Bounds will be entitled to receive (i) a cash severance payment equal to \$1,164,500, less applicable withholdings and taxes, payable over a six-month period, (ii) continued group health plan coverage for Mr. Bounds, his eligible spouse and other dependents for a period of up to one year following the Separation Date, provided that Mr. Bounds, his spouse and other dependents elect to maintain group health plan coverage for such period, and (iii) \$95,500 for attorneys' fees. In addition, Mr. Bounds will be paid his base salary through the Separation Date. The Separation Agreement provides that the receipt of any benefits pursuant to the Separation Agreement is contingent upon the execution of a waiver and mutual release of claims on the Separation Date. Mr. Bounds' departure is not related to any issues regarding disagreements with management or the Board.

The Separation Agreement also provides for certain customary covenants regarding confidentiality and non-disparagement. Upon effectiveness of the Separation Agreement, Mr. Bounds' existing employment agreement will automatically terminate; provided, however, that certain surviving customary confidentiality provisions and restrictive covenants will remain in full force and effect.

The foregoing description of the Separation Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of such document, which is filed as Exhibit 10.1 to this Current Report on Form 8-K and incorporated by reference herein.

Appointment of New Chief Executive Officer

On April 9, 2021, the Board appointed Mr. George Maxwell as Chief Executive Officer of the Company, effective April 19, 2021. Concurrent with the appointment of Mr. Maxwell, the Company and Mr. Maxwell entered into an employment agreement effective as of April 19, 2021 (the "*Employment Agreement*"), pursuant to which Mr. Maxwell will be entitled to receive an annual base salary of \$450,000. The Employment Agreement also provides that Mr. Maxwell will be eligible to receive an annual cash bonus with a target percentage equal to 50% of his base salary and stock options and other long-term incentive awards up to 50% of his base salary. Pursuant to the Employment Agreement, the Company will pay Mr. Maxwell \$22,000 per year for health benefits and \$17,000 per year for pension benefits, as well as provide other customary employment benefits including paid vacation and sick leave.

The foregoing description of the Employment Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of such document, which is filed as Exhibit 10.2 to this Current Report on Form 8-K and incorporated by reference herein.

Mr. Maxwell founded Eland Oil & Gas Plc. in 2009 and served as the company's Chief Executive Officer from September 2014 to December 2019, Chief Financial Officer from 2010 to 2014, and as a member of the board of directors from 2009 to 2019, until the company was acquired by Seplat Petroleum Development Company Pls. on December 17, 2019. Prior to founding Eland Oil & Gas Plc., Mr. Maxwell served as the business development manager for Addax Petroleum and, prior to this, commercial manager in Geneva. Mr. Maxwell joined Addax Petroleum in 2004 and held the general manager position in Nigeria, where he was responsible for finance, and fiscal and commercial activities. Prior to this, Mr. Maxwell worked with ABB Oil & Gas as vice president of finance based in the United Kingdom with responsibilities for Europe and Africa. He held a similar position in Houston, from where the organization ran its operations in ten countries. Mr. Maxwell was finance director in Singapore for Asia Pacific and Middle East, handling currency swaps and minimizing exposures during the Asian financial crisis of the late 1990s. Mr. Maxwell graduated from Robert Gordon University in Aberdeen with a Master's in Business Administration. Mr. Maxwell is a Fellow of the Energy Institute in the UK and has formerly served on the boards of directors of Elcrest Exploration and Production Nigeria Ltd., and Westport Oil Limited.

There is no family relationship between Mr. Maxwell and any director or executive officer of the Company. There are no transactions between Mr. Maxwell and the Company that would be required to be reported under Item 404(a) of Regulation S-K of the Securities Exchange Act of 1934, as amended.

Item 7.01 Regulation FD Disclosure.

On April 12, 2021, the Company issued a press release announcing the appointment of Mr. Maxwell and the resignation of Mr. Bounds. A copy of the press release is attached hereto as Exhibit 99.1.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description
10.1	Separation Agreement, dated April 9, 2021
10.2	Employment Agreement, effective as of April 19, 2021
99.1	Press Release, dated April 12, 2021
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

VAALCO ENERGY, INC. (Registrant)

Date: April 12, 2021

By: /s/ Jason Doornik

Name: Jason Doornik Title: Chief Accounting Officer and Controller



VAALCO Energy, Inc. 9800 Richmond Ave., Suite 700 Houston, Texas 77042 *Tel: (713) 623-0801*

SEPARATION AND MUTUAL RELEASE AGREEMENT

This **SEPARATION AND MUTUAL RELEASE AGREEMENT** (the "*Agreement*") is made and entered into as of April 9, 2021, by and between VAALCO Energy, Inc. (the "*Company*") and Cary Bounds ("*Executive*"). The Company and Executive may be referred to singularly as a "*Party*" or collectively as the "*Parties*." Defined terms not defined in this Agreement have the meaning set forth in the Employment Agreement (as defined below).

WHEREAS, Executive has been employed by the Company as its Chief Executive Officer ("CEO") and Chief Operating Officer ("COO") and is a member of the Company's Board of Directors (the "Board");

WHEREAS, Executive executed an Employment Agreement with the Company effective as of December 29, 2016 ('Employment Agreement'), which, except for the Surviving Covenants, shall terminate as of the Separation Date;

WHEREAS, the Parties have mutually agreed that Executive's employment with the Company shall end as of 5 p.m. CST on April 30, 2021 (the "Separation Date"); and

WHEREAS, the Parties desire to set forth Executive's separation benefits and obligations and to finally, fully and completely resolve all matters arising from or during Executive's employment and separation from employment, any benefits, bonuses and compensation connected with such employment and all other disputes and matters that the Parties may have for any reason; and

THEREFORE, in consideration of the mutual representations contained herein, and such other good and valuable consideration, the sufficiency of which Executive acknowledges, the Parties hereby covenant and agree, with the intent to be legally bound, as follows:

1. <u>Transition</u>. During the period beginning on the Effective Date and ending on the Separation Date, (the 'Remaining Employment Term'), the Parties agree that they each shall continue to perform their respective obligations pursuant to the terms and conditions of the Employment Agreement (including, without limitation, payment of Executive's base salary during such Remaining Employment Term), provided, however, that immediately upon the Effective Date, Executive agrees that by virtue of this Agreement he resigns as the Company's CEO and COO and as a member of the Board (and any committees thereof) and a member of the board of directors or officer of any subsidiary of the Company on April 18, 2021, and throughout the Remaining Employment Term, Executive agrees that he shall solely be responsible for assisting with the transition of his duties and responsibilities to such persons as the Company may designate so as to cause an orderly transition of such duties and responsibilities. Executive agrees of the Company term the Separation Date will be subject to the same standards of conduct and performance applicable to all officers and employees of the Company taking into account his position and duties during the Remaining Employment Term.

SEPARATION AND MUTUAL RELEASE AGREEMENT

- 2. <u>Consideration</u>. Subject to the terms and conditions of this Agreement, the Company agrees to pay to Executive the amounts described in subparagraphs (a) (c) of this <u>Section 2</u> (collectively, the "*Separation Benefits*"), provided Executive signs the Form of Waiver and Mutual Release of Claims attached hereto as <u>Exhibit A</u> (the "*Release*") on (but not before) the Separation Date (or, if later, before the end of the Release Review Period as defined in the Release) and does not revoke the Release in accordance with its terms, in consideration for Executive's promises herein, including, without limitation, the Release:
 - (a) A severance payment equal to \$1,164,500, payable to Executive over asix-month period in 12, substantially equal bi-monthly payments that begin on the Company's first payroll day following the Release Effective Date (as defined in the Release), less applicable withholdings and taxes;
 - (b) Continued group health plan coverage following the Separation Date under any of the Company's group health plans that covered Executive immediately before the Separation Date that are subject to the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, as codified in Code Section 4980B and Part 6 of Subtitle B of Title I of ERISA ("COBRA"), for Executive and his eligible spouse and other dependents (together, "Dependents"), for a period of one year following the Separation Date and at no cost to Executive and his Dependent; provided that Executive and his Dependents, if any, elect COBRA continuation coverage and maintain such coverage for such one year period; and
 - (c) The amount of \$95,500 as MISC 1099 Income for attorneys' fees, which the Company agrees to pay directly to Executive's counsel as soon as administratively practicable, and in no event later than 30 days following the later of the Release Effective Date, provided that the attorney provides attorney's completed Form W-9 to the Company.

Executive confirms and agrees that he would not otherwise have received, or been entitled to receive, the Separation Benefits if he did not enter into this Agreement. The Company shall also pay Executive the Minimum Payments set forth in <u>Article 4, Section 4.1(a)(i), (ii) and (iii)</u> of the Employment Agreement. All other compensation and benefits provided by the Company shall cease upon the Separation Date, unless otherwise expressly provided for by the express terms of such benefits plans or programs or this Agreement.

3. Termination of Employment. The Parties agree that Executive's employment with the Company and all of Executive's positions with the Company and all Affiliates shall terminate effective as of the Separation Date. Executive shall immediately execute all documents and take such further steps as may be required to effectuate such termination(s). All written employment agreements shall terminate as of the Separation Date, including without limitation, the Employment Agreement <u>except for</u> the Surviving Covenants (defined below) which are not extinguished by this Agreement and are incorporated into this Agreement by reference. Executive and the Company agree that the following provisions in the Employment Agreement survive the termination of the Employment Agreement and the termination of Executive's employment and shall remain in full force and effect as set forth in the Employment Agreement: <u>Confidential Information and Restrictive Covenants</u> in <u>Article 4</u> and the <u>General Provisions</u> in <u>Article 5</u> of the Employment Agreement (collectively, the "*Surviving Covenants*"). For purposes of this Agreement: (a) the term "*Affiliate*" means any person or entity, controlling, controlled by, or under common control with the Company, and (b) the terms "*controlling, controlled by, or under common control*" mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies (whether through ownership of securities or any partnership or other ownership interest, by contract or otherwise) of another person or entity. Whether any person or entity is an Affiliate will be determined solely by the Company.

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 2 of 11

- 4. <u>No Other Payments</u>. Executive understands and agrees that the Company shall make no other payments to Executive, other than the consideration set forth in this Agreement, and shall have no other obligations to Executive except as described in this Agreement. Executive acknowledges that Executive has no right to seek, and will not seek, any additional or different compensation or consideration for executing or performing under this Agreement. Further, the Parties acknowledge and agree that the provisions of Section 4.1(b) of the Employment Agreement relating to a CIC Severance Payment Event shall no longer apply after the Separation Date.
- 5. <u>Cooperation</u>. As a further material inducement to the Company to provide Executive the Separation Benefits, Executive agrees (A) to cooperate and provide reasonable assistance, at the request of the Company, in the transitioning of Executive's job duties and responsibilities, (B) to be reasonably available to the Company or its representatives (including attorneys) to provide general advice or assistance as requested by the Company, and (C) to cooperate and provide reasonable assistance, at the request of the Company, in any and all investigations or other legal, equitable or business matters or proceedings which involve any matters for which Executive worked on or had responsibility during Executive's employment with the Company. This includes but is not limited to testifying (and preparing to testify) as a witness in any proceeding or otherwise providing information or reasonable assistance to the Company in connection with any investigation, claim or suit, and cooperating with the Company regarding any investigation, claims or other disputed items involving the Company that relate to matters within the knowledge or responsibility of Executive shall receive no additional compensation for time spent assisting the Company pursuant to this <u>Section 5</u> other than the compensation and benefits provided for in this Agreement, provided that Executive shall be entitled to be reimbursed for any reasonable out-of-pocket expenses incurred in fulfilling Executive's obligations pursuant to this <u>Section 5</u>.

6. <u>Mutual Release of Claims</u>.

Release of Claims by Executive. In consideration of the promises of the Company provided herein, including, the consideration provided for (a) in Section 2 of this Agreement, the Company's release of claims and other consideration provided for in this Agreement, that being good and valuable consideration, the receipt, adequacy and sufficiency of which Executive acknowledges, Executive, on Executive's own behalf and on behalf of Executive's agents, administrators, representatives, executors, successors, heirs, devisees and assigns (collectively, the "Executive Releasing Parties") hereby fully and forever waives, releases, extinguishes and discharges the Company and its, Affiliates, parents and subsidiaries, present and future parents, owners, officers, directors, stockholders, members, executives, employees, consultants, independent contractors, partners, agents, attorneys, advisers, insurers, fiduciaries, employee benefit plans, representatives, successors and assigns (each, a "Company Released Party" and collectively, the "Company Released Parties"), jointly and severally, from any and all claims, rights, demands, debts, obligations, losses, causes of action, suits, controversies, setoffs, affirmative defenses, counterclaims, third party actions, damages, penalties, costs, expenses, attorneys' fees, liabilities and indemnities of any kind or nature whatsoever (collectively, the "Claims"), whether known or unknown, suspected or unsuspected, accrued or unaccrued, whether at law, equity, administrative, statutory or otherwise, and whether for injunctive relief, back pay, front pay, fringe benefits, equity, reinstatement, reemployment, compensatory damages, punitive damages, or any other kind of damages, which any of Executive Releasing Parties have, had or may have against any of the Company Released Parties relating to or arising out of any matter arising on or before the date this Agreement is executed by Executive. Such released Claims include, without limitation, all Claims

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 3 of 11

arising from or relating to Executive's employment with the Company or the termination of that employment relationship or any circumstances related thereto, or any other agreement, matter, cause or thing whatsoever, including without limitation all Claims arising under or relating to Executive's employment, any profit sharing or profits interest plan or arrangement, equity, compensation, bonuses, benefits, payments, or any other benefits or payments Executive may or may not have received during Executive's employment with the Company, all Claims relating to any other claimed payments, employment contracts or benefits, all Claims arising from or relating to Executive's performance of services for the Company and any of its Affiliates during Executive's employment with the Company, including without limitation all Claims arising at law or equity or sounding in contract (express or implied) or tort, Claims arising by statute, common law or otherwise, Claims arising under any federal, state, county or local laws, of any jurisdiction, including Claims for wrongful discharge, libel, slander, breach of express or implied contract or implied covenant of good faith and fair dealing, Claims for alleged fraud, concealment, negligence, negligent misrepresentation, promissory estoppel, quantum meruit, intentional or negligent infliction of emotional distress, violation of public policy, and Claims for discrimination, retaliation, sexual harassment and Claims arising under any laws that prohibit age, sex, sexual orientation, race, national origin, color, disability, religion, veteran, workers' compensation or any other form of discrimination, harassment, or retaliation, including, without limitation, Claims under the Age Discrimination in Employment Act of 1967, as amended, the Americans with Disabilities Act of 1990, as amended, the Rehabilitation Act of 1973, Title VII of the Civil Rights Act of 1964, as amended, 42 U.S.C. §1981, the Civil Rights Act of 1991, the Civil Rights Act of 1866 and/or 1871, the Equal Pay Act of 1963, the Lilly Ledbetter Fair Pay Act of 2009, the Fair Labor Standards Act, the Employee Retirement Income Security Act of 1974, as amended, the Family and Medical Leave Act of 1993, the Occupational Safety and Health Act, the Employee Polygraph Protection Act, the Uniformed Services Employment and Reemployment Rights Act, the Worker Adjustment and Retraining Notification Act, the Genetic Information Nondiscrimination Act, the National Labor Relations Act, the Labor Management Relations Act, the Immigration Reform and Control Act, the Texas Labor Code, the Texas Payday Law, the Texas Commission on Human Rights or Chapter 21, any statute or laws of the State of Texas or any other federal, state, local, municipal or common law whistleblower, discrimination or anti-retaliation statute law or ordinance, and any other Claims arising under state, federal, local, municipal or common law, as well as any expenses, costs or attorneys' fees.

Except as required by law, Executive agrees that Executive will not commence, maintain, initiate, or prosecute, or cause, encourage, assist, volunteer, advise or cooperate with any other person to commence, maintain, initiate or prosecute, any action, lawsuit, proceeding, charge, petition, complaint or Claim before any court, agency or tribunal against the Company or any of the Company Released Parties arising from, concerned with, or otherwise relating to, in whole or in part, Executive's employment, the terms and conditions of Executive's employment, or Executive's separation from employment with the Company or any of the matters or Claims discharged and released in this Agreement. Executive represents that Executive has not filed any complaints, charges or lawsuits against the Company with any governmental agency or any curt based on Claims that are released and waived by this Agreement.

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 4 of 11

- (b) By the Company. In consideration of the mutual promises contained in this Agreement, including Executive's release of claims, which is in addition to anything of value to which the Company is already entitled, the Company, on behalf of itself and all of its parents, divisions, subsidiaries, Affiliates, joint venture partners, partners, and related companies, and their present and former agents, executives, employees, officers, directors, attorneys, stockholders, plan fiduciaries, successors and assigns, irrevocably and unconditionally releases, waives, and forever discharges, Executive and his heirs, executors, successors and assigns (the "*Executive Released Parties*"), from any and all claims, demands, actions, causes of action, costs, fees, and all liability whatsoever, whether known or unknown, fixed or contingent, which the Company has, had, or may have against the Executive Released Parties relating to or arising out of his employment, compensation and terms and conditions of employment or separation from employment up through the Effective Date. This release includes, without limitation, claims at law or equity or sounding in contract (express or implied) or tort, claims arising under any federal, state or local laws, or any other statutory or common law claims related to relating to or arising out of Executive's employment or separation from employment for any period up to and including the date this Agreement is signed by the Parties.
- 7. Exclusions from Mutual Release. Notwithstanding any other provision of this Agreement to the contrary:
 - (a) This Agreement specifically does not release any claim or cause of action by or on behalf of Executive (or his beneficiary) (i) for any payment or other benefit that is required under this Agreement or pursuant to any Plan (as defined in the Employment Agreement prior to the receipt thereof by or on behalf of Executive), (ii) arising out of the Company's obligation to indemnify Executive in his capacity as a director, officer or employee of the Company or any Affiliate thereof, or as a former director, officer or employee of the Company's by-laws, any agreement to which Executive is a party or beneficiary, at law, or otherwise, or (iii) arising after the Effective Release Date.
 - (b) The Parties may report possible violations of federal, state or local law or regulation to any governmental or law enforcement agency or entity or may disclose Confidential Information when required to do so by a court of competent jurisdiction, by any governmental agency having authority over Executive or the business of the Company or by any administrative body or legislative body (including a committee thereof) with jurisdiction to order Executive or the Company to divulge, disclose or make accessible such information. Nothing in this Agreement is intended to interfere a Party's right to (i) report possible violations of state or federal law or regulation to any governmental agency or entity, (ii) make other disclosures that are protected under the whistleblower provisions of state or federal law or regulation, (iii) file a claim or charge with the U.S. Equal Employment Opportunity Commission ("EEOC"), any state human rights commission, or any other government agency or entity, or (iv) testify, assist, or participate in an investigation, hearing, or proceeding conducted by the EEOC, any state human rights commission, or any other government or law enforcement agency, entity or court. However, Executive understands and agrees that if Executive, or someone acting on Executive's behalf, should file, or cause to be filed, any claim, charge, complaint, or action against the Company and/or any other Company Released Parties subject to the release of claims in Section 6, Executive expressly waives any and all rights to recover any damages or other relief from the Company and/or any other Company Released Parties including, without limitation, damages, costs and attorneys' fees. Similarly, by executing this Agreement, Executive hereby waives the right to recover any damages or benefits in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other government agency or entity or in any proceeding brought by the EEOC, any state human rights commission, or any other government agency or entity on Executive's behalf;

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 5 of 11

- (c) Executive is hereby notified in accordance with the Defend Trade Secrets Act of 2016 that Executive will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that is made in confidence to a federal, state, or local government official, either directly or indirectly, or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, or is made in a complaint or other document that is filed under seal in a lawsuit or other proceeding. If Executive files a lawsuit for retaliation against the Company for reporting a suspected violation of law, Executive may disclose the Company's trade secrets to Executive's attorney and use the trade secret information in the court proceeding if Executive files any document containing the trade secret under seal, and does not disclose the trade secret, except pursuant to court order;
- (d) This Agreement does not release any claim that may not be waived by applicable law; and
- (e) This Agreement does not release any claim relating to the Parties' obligations under this Agreement, the Parties' rights to enforce any of the Surviving Covenants; and any claims arising after the Effective Release Date.

8. <u>Mutual Non-Disparagement</u>.

- (a) Executive agrees that the Company's goodwill and reputation are assets of great value to the Company which have been obtained and maintained through great costs, time and effort. Therefore, Executive hereby agrees not to disclose, communicate, or publish any disparaging or negative information, writings, electronic communications, comments, opinions, facts, or remarks, of any kind, about the Company Released Parties; provided, however, that this paragraph shall have no application to any evidence or testimony required by any court or other government entity, including but not limited to, the EEOC or any similar federal, state or local agency, pursuant to compulsion of law. Executive acknowledges that in executing this Agreement, Executive has knowingly, voluntarily and intelligently waived any free speech or First Amendment rights under the United States Constitution or applicable state counterpart to disclose, publish or communicate any such disparaging information about the Company and/or any of the other Company Released Parties.
- (b) The Company agrees that Executive's goodwill and reputation are assets of great value to Executive which have been obtained and maintained through great costs, time and effort. Therefore, the Company's officers and the members of the Board shall not disclose, communicate, or publish any disparaging or negative information, writings, electronic communications, comments, opinions, facts, or remarks, of any kind, about the Executive.
- (c) The rights afforded the Parties under this <u>Section 8</u> are in addition to any and all rights and remedies otherwise afforded by law. This <u>Section 8</u> shall not preclude the Parties from providing truthful testimony in a court of law or a government investigation, nor shall this <u>Section 8</u> shall interfere with Executive's rights in <u>Section 7(b)</u>.

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 6 of 11

- 9. <u>Public Announcements</u>. The Company agrees to consult with Executive regarding any press release or public announcement, before any such press release or public announcement is made, and Executive agrees not to speak with the press or make any press release or make any public announcement without the prior written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed) regarding his separation from employment, provided, however, that the foregoing shall not prohibit any communications that are required by applicable law, by any governmental or regulatory or self-regulatory agency, or that are a part of normal reporting, rating or review procedures or fundraising, marketing, information or reporting activities of any stockholder of the Company.
- 10. Known Violations. Executive represents and warrants that Executive is not aware of any illegal acts committed by or on behalf of the Company and represents that if Executive is or had been aware of any such conduct, that Executive has properly reported the same to the Board or the Company's Vice President and General Counsel in writing. Executive further represents and warrants that Executive is not aware of any (i) violations, allegations or claims that the Company has violated any federal, state, local or foreign law or regulation of any kind, or (ii) any facts, basis or circumstances relating to any alleged violations, allegations or claims that the Company has violated any federal, state the Company has violated any federal, state, local or foreign law or regulation of any kind. If Executive learns of any such information, Executive shall immediately inform the Company's General Counsel.
- 11. <u>No Admission of Liability</u>. Executive understands and agrees that this Agreement shall not in any way be construed as an admission by the Company, or by any of the other Company Released Parties, of any unlawful or wrongful acts whatsoever against Executive or any other person. The Company Released Parties specifically disclaim any liability to, or wrongful acts against, Executive or any other person.
- 12. Duty to Return Company Property. On the Separation Date, Executive will immediately return to the Company all property of the Company or its Affiliates, including, without limitation, Confidential Information (as defined in the Employment Agreement), keys, equipment, computer and computer equipment, devices, cellular phones, credit cards, data, lists, information, correspondence, notes, memos, reports, or other writings prepared by the Company or any of its Affiliates or by Executive on behalf of the Company or any of its Affiliates. Executive agrees that Executive shall not take, copy, alter, destroy, or delete any files, documents or other materials whether or not embodying or recording any Confidential Information, including copies, without obtaining the advance written consent of Michael Silver, Vice President and General Counsel.
- 13. <u>Remedies</u>. The Parties understand and agree that if a violation of term of this Agreement is asserted, the Party who asserts such violation shall have the right to seek specific performance of that term and/or any other necessary and proper relief as permitted by law or equity, including but not limited to, damages awarded by any arbitrator or court of competent jurisdiction, and the prevailing party shall be entitled to recover its reasonable costs and attorneys' fees.

In the event Executive breaches or threatens to breach any portion, or challenges the enforceability, of this Agreement or the Surviving Covenants, Executive shall forfeit all remaining but unpaid Separation Benefits from such date, and shall pay the Company (i) an amount equal to the Separation Benefits paid to Executive from the date of such breach or threatened breach, (ii) all attorneys' fees, expenses and costs the Company incurs in such action, and (iii) any and all other damages to which the Company may be entitled at law or in equity as a result of a breach of this Agreement or the Surviving Covenants.

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 7 of 11

- 14. Knowing and Voluntary Waiver of Age Claims; Time to Consider Offer of Separation Benefits. Executive, by Executive's free and voluntary act of signing below, acknowledges that (a) Executive has been given a period of 21 days ("Review Period") to consider whether to agree to the terms contained in this Agreement, (b) the Company is advising and has advised Executive in writing (*i.e.*, through this Agreement) to consult with an attorney of Executive's own choosing at Executive's cost, regarding the effect of this Agreement, and Executive has had a reasonable opportunity to do so, if so desired, (c) Executive understands that this Agreement specifically releases and waives all rights and claims Executive may have under the Age Discrimination in Employment Act ("ADEA") prior to the date on which Executive signs this Agreement, and (d) Executive knowingly and voluntarily agrees to all of the terms of this Agreement and intends to be legally bound thereby. The Parties agree that any changes to this Agreement, whether material or immaterial, will not restart the running of the Review Period. If Executive does not sign this Agreement before the Review Period expires, this Agreement offer will be withdrawn automatically. Executive understands that this Review Period may be waived and Executive agrees that the decision to do so was voluntary and that he took the reasonable time necessary to consult with counsel. The procedure for Executive to accept this Agreement is to return a fully executed, dated, and witnessed Agreement to Michael Silver, Vice President and General Counsel, prior to the last day of the Review Period.
- 15. <u>Seven Day Revocation Period</u>. Executive may revoke this Agreement at any time within seven days after Executive returns a signed copy pursuant to <u>Section 14</u>. To revoke the Agreement, Executive must deliver written notification of such revocation to the attention of Michael Silver, Vice President and General Counsel, within seven days after the date that Executive signs this Agreement. Executive understands that if Executive does not revoke the Agreement within seven days following execution (excluding the date of execution), the Agreement will become effective and binding on the Parties, and fully enforceable by the Parties, as of the date of execution (the "*Effective Date*").
- 16. <u>Severability</u>. The Parties fully intend that this Agreement comply with all applicable laws and legal requirements. Should any provision of this Agreement be declared or be determined by any court of competent jurisdiction to be illegal, invalid or otherwise unenforceable, the Agreement shall first be reformed to make the provision at issue enforceable and effective to the fullest extent permitted by law. Further, if a court should determine that any portion of this Agreement is unenforceable, such provision shall be given effect to the maximum extent possible by narrowing or enforcing in part that aspect of the provision found to be unenforceable. If such reformation is not possible, all remaining provisions of this Agreement shall otherwise remain in full force and effect and shall be construed as if such illegal, invalid, or unenforceable provision had not been included herein.
- 17. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the Parties, and their respective heirs, executors, beneficiaries, personal representatives, successors and permitted assigns hereunder, but otherwise this Agreement shall not be for the benefit of any third parties.
- 18. Entire Agreement; Amendment. This Agreement sets forth the entire agreement of the Parties and fully supersedes and replaces any and all prior agreements, promises, representations, or understandings, written or oral, between the Company (and any other Company Released Party) and Executive that relates to the subject matter of this Agreement, unless referenced in this Agreement and, therefore, incorporated into this Agreement by reference (including the Surviving Covenants), other than any equity award agreements between Executive and the Company outstanding on the Separation Date (which awards shall continue to be governed by their applicable

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 8 of 11

award agreements). Executive acknowledges that in executing this Agreement, (a) Executive does not rely, and has not relied, upon any oral or written representation, promise or inducement by the Company and/or any of the other Company Released Parties, except as expressly contained in this Agreement, (b) Executive has relied solely on his own judgment (and if applicable, that of his own counsel), in entering into this Agreement, and (c) there is no presumption regarding the interpretation or construction of this Agreement against its drafter. Executive understands and agrees that Executive is precluded from bringing any fraud or similar claim against the Company or any of the other Company Released Parties associated with any such communications, representations, promises or inducements. This Agreement may be amended or modified only by a written instrument identified as an amendment hereto that is executed by both Parties.

- 19. <u>Survival of Certain Provisions</u>. Wherever appropriate to the intention of the Parties, the respective rights and obligations of the Parties hereunder shall survive any termination or expiration of this Agreement.
- 20. <u>Code Section 409A</u>. It is intended that this Agreement shall either be exempt from or, to the extent it is not exempt from, comply with the provisions of Section 409A of the Internal Revenue Code of 1986, as amended ("*Code Section 409A*") and the regulations and guidance of general applicability issued thereunder so as to not subject the Executive to the payment of additional interest and taxes under Code Section 409A, and in furtherance of this intent, this Agreement shall be interpreted, operated, and administered in a manner consistent with these intentions. Any installment payment to be made pursuant to this Agreement shall be treated as a separate payment for purposes of Code Section 409A to the extent Code Section 409A applies to such payments.
- 21. Choice of Law and Forum. This Agreement shall be governed by, and construed and interpreted in accordance with, the laws of the State of Texas, but without regard to principles of conflict of laws that might direct the application of the law of another forum. Any claim or dispute between the Parties shall be governed by the arbitration provision in the General Provisions of the Employment Agreement. To the extent not covered by such arbitration provision, any action to enforce the provisions of this Agreement, or otherwise relating to this Agreement, must be brought in any court of competent jurisdiction in Harris County, Texas, and the Parties hereby waive any objection to such venue including, without limitation, that it is inconvenient for either Party.
- 22. <u>Waiver of Jury Trial</u>. THE PARTIES WAIVE ANY RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THE AGREEMENT OR ANY CLAIM HEREUNDER, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES TO IRREVOCABLY WAIVE TRIAL BY JURY, AND THAT ANY PROCEEDING WHATSOEVER BETWEEN THEM RELATING TO THE AGREEMENT OR ANY CLAIM HEREUNDER SHALL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.
- 23. <u>Waiver</u>. A Party's waiver of any breach or violation of any provision of this Agreement shall not operate as, or be construed to be, a waiver of any later breach of the same or other provision by such Party.
- 24. <u>Counterparts</u>. The Parties agree that the Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall be deemed one and the same instrument.

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 9 of 11

PLEASE READ CAREFULLY BEFORE SIGNING

- Executive acknowledges that he has carefully read and understands the terms of this Agreement and all of Executive's promises and obligations hereunder.
- Executive acknowledges that he has been advised to review this Agreement with an attorney before signing it.
- Executive understands that he is releasing all claims against the Company and all Affiliates (except as set forth in this Agreement), including claims under the Age Discrimination in Employment Act.
- Executive acknowledges that he has been given at least 21 days to consider whether to sign this Agreement. Executive acknowledges that if he signs this Agreement before the end of the 21-day period, it will be Executive's own personal and voluntary decision to do so.
- Executive understands that this Agreement will not become effective or enforceable until after the7-day revocation period has expired. The Company will have no obligations to Executive under this Agreement if Executive revokes the Agreement during such 7-day period.

[Signature page follows]

SEPARATION AND MUTUAL RELEASE AGREEMENT

Page 10 of 11

Please review this document carefully as it includes a release of claims.

IN WITNESS WHEREOF, Executive has entered into this Agreement, and the Company has caused this Agreement to be executed in its name and on its behalf by its duly authorized officer.

EXECUTIVE

/s/ Cary Bounds Cary Bounds

Date: 04/09/2021

VAALCO ENERGY, INC.

By: <u>/s/ Andrew Fawthrop</u> Printed Name: Andrew Fawthrop Title: Chairman of the Board of Directors Date: 04/09/2021

Address for notices:

VAALCO Energy, Inc. 9800 Richmond Ave. Suite 700 Houston, TX 77042 Attn: Michael Silver Vice President and General Counsel WITNESS

/s/ Erin Bounds Witness Signature

Title: _____ Printed Name: Erin Bounds Date: 04/09/2021

Separation Agreement Signature Page

EXHIBIT A

MUTUAL RELEASE OF CLAIMS

VAALCO Energy, Inc. (the "*Company*") and Cary Bounds *Executive*"), enter into this Mutual Release of Claims effective as set forth below in <u>Section 2</u>. Terms used in this Release with initial capital letters that are not otherwise defined herein shall have the meanings ascribed to such terms in the Separation and Mutual Release Agreement entered into on April 9, 2021, by and between the Company and Executive (the "*Agreement*").

WHEREAS, Executive and the Company are parties to the Agreement; and

WHEREAS, Section 2 of the Agreement provides that Executive is entitled to certain payments and benefits if he signs this Release.

NOW THEREFORE, in consideration of the mutual promises and covenants set forth herein, the receipt and adequacy of which are acknowledged, Executive and the Company agree as follows:

1. Mutual Release of Claims.

(a) Release of Claims by Executive. In consideration of the promises of the Company provided herein, including, the consideration provided for in Section 2 of the Agreement and other consideration provided for in the Agreement and the Company's release of claims, that being good and valuable consideration, the receipt, adequacy and sufficiency of which Executive acknowledges, Executive, on Executive's own behalf and on behalf of Executive's agents, administrators, representatives, executors, successors, heirs, devisees and assigns (collectively, the "Executive Releasing Parties") hereby fully and forever waives, releases, extinguishes and discharges the Company, its Affiliates, its or their stockholders, subsidiaries and each of their respective past, present and future parents, owners, officers, directors, stockholders, members, executives, employees, consultants, independent contractors, partners, agents, attorneys, advisers, insurers, fiduciaries, employee benefit plans, representatives, successors and assigns (each, a "Company Released Party" and collectively, the "Company Released Parties"), jointly and severally, from any and all claims, rights, demands, debts, obligations, losses, causes of action, suits, controversies, setoffs, affirmative defenses, counterclaims, third party actions, damages, penalties, costs, expenses, attorneys' fees, liabilities and indemnities of any kind or nature whatsoever (collectively, the "Claims"), whether known or unknown, suspected or unsuspected, accrued or unaccrued, whether at law, equity, administrative, statutory or otherwise, and whether for injunctive relief, back pay, front pay, fringe benefits, equity, reinstatement, reemployment, compensatory damages, punitive damages, or any other kind of damages, which any of Executive Releasing Parties have, had or may have against any of the Company Released Parties relating to or arising out of any matter arising on or before the date this Release is executed by Executive. Such released Claims include, without limitation, all Claims arising from or relating to Executive's employment with the Company or the termination of that employment relationship or any circumstances related thereto, or any other agreement, matter, cause or thing whatsoever, including without limitation all Claims arising under or relating to Executive's employment, any profit sharing or profits interest plan or arrangement, equity, compensation, bonuses, benefits, payments, or any other benefits or payments Executive may or may not have received during Executive's employment with the Company, all Claims relating to any other claimed payments, employment contracts or benefits, all Claims arising from or relating to Executive's performance of services for the Company and any of its Affiliates during Executive's employment with the Company, including without limitation all Claims arising at law or equity or sounding in contract (express or implied) or tort, Claims arising by statute, common law or otherwise, Claims arising under any federal, state, county or local laws, of any jurisdiction, including Claims for

RELEASE OF CLAIMS

Page 1 of 6

wrongful discharge, libel, slander, breach of express or implied contract or implied covenant of good faith and fair dealing, Claims for alleged fraud, concealment, negligence, negligent misrepresentation, promissory estoppel, quantum meruit, intentional or negligent infliction of emotional distress, violation of public policy, and Claims for discrimination, retaliation, sexual harassment and Claims arising under any laws that prohibit age, sex, sexual orientation, race, national origin, color, disability, religion, veteran, workers' compensation or any other form of discrimination, harassment, or retaliation, including, without limitation, Claims under the Age Discrimination in Employment Act of 1967, as amended, the Americans with Disabilities Act of 1990, as amended, the Rehabilitation Act of 1973, Title VII of the Civil Rights Act of 1964, as amended, 42 U.S.C. §1981, the Civil Rights Act of 1991, the Civil Rights Act of 1866 and/or 1871, the Equal Pay Act of 1963, the Lilly Ledbetter Fair Pay Act of 2009, the Fair Labor Standards Act, the Employee Retirement Income Security Act of 1974, as amended, the Family and Medical Leave Act of 1993, the Occupational Safety and Health Act, the Employee Polygraph Protection Act, the Uniformed Services Employment and Reemployment Rights Act, the Worker Adjustment and Retraining Notification Act, the Genetic Information Nondiscrimination Act, the National Labor Relations Act, the Labor Management Relations Act, the Immigration Reform and Control Act, the Texas Labor Code, the Texas Payday Law, the Texas Commission on Human Rights or Chapter 21, any statute or laws of the State of Texas or any other federal, state, local, municipal or common law whistleblower, discrimination or anti-retaliation statute law or ordinance, and any other Claims arising under state, federal, local, municipal or common law, as well as any expenses, costs or attorneys' fees. Except as required by law, Executive agrees that Executive will not commence, maintain, initiate, or prosecute, or cause, encourage, assist, volunteer, advise or cooperate with any other person to commence, maintain, initiate or prosecute, any action, lawsuit, proceeding, charge, petition, complaint or Claim before any court, agency or tribunal against the Company or any of the Company Released Parties arising from, concerned with, or otherwise relating to, in whole or in part, Executive's employment, the terms and conditions of Executive's employment, or Executive's separation from employment with the Company or any of the matters or Claims discharged and released in this Release. Executive represents that Executive has not filed any complaints, charges or lawsuits against the Company with any governmental agency or any court based on Claims that are released and waived by this Release.

(b) By the Company. In consideration of the mutual promises contained in this Release, including Executive's release of claims, which is in addition to anything of value to which the Company is already entitled, the Company, on behalf of itself and all of its parents, divisions, subsidiaries, affiliates, joint venture partners, partners, and related companies, and their present and former agents, executives, employees, officers, directors, attorneys, stockholders, plan fiduciaries, successors and assigns, irrevocably and unconditionally releases, waives, and forever discharges, Executive and his heirs, executors, successors and assigns (the "*Executive Released Parties*"), from any and all claims, demands, actions, causes of action, costs, fees, and all liability whatsoever, whether known or unknown, fixed or contingent, which the Company has, had, or may have against the Executive Released Parties relating to or arising out of his employment, compensation and terms and conditions of employment or separation from employment up through the Release Effective Date. This release includes, without limitation, claims at law or equity or sounding in contract (express or implied) or tort, claims arising under any federal, state or local laws, or any other statutory or common law claims related to relating to or arising out of Executive's employment or separation from employment for any period up to and including the date this Release is signed by the Parties.

(c) Claims Excluded from Release. Notwithstanding any other provision of this Release to the contrary:

RELEASE OF CLAIMS

Page 2 of 6

(i) This Release specifically does not release any claim or cause of action by or on behalf of Executive (or his beneficiary) (i) for any payment or other benefit that is required under this Release or pursuant to any Plan (as defined in the Employment Agreement prior to the receipt thereof by or on behalf of Executive) or (ii) arising out of the Company's obligation to indemnify Executive in his capacity as a director, officer or employee of the Company or any Affiliate thereof, or as a former director, officer or employee of the Company or any Affiliate as provided in the Company's by-laws, any agreement to which Executive is a party or beneficiary, at law, or otherwise, or (iii) arising after the Effective Release Date;

(ii) The Parties may report possible violations of federal, state or local law or regulation to any governmental or law enforcement agency or entity or may disclose Confidential Information when required to do so by a court of competent jurisdiction, by any governmental agency having authority over Executive or the business of the Company or by any administrative body or legislative body (including a committee thereof) with jurisdiction to order Executive or the Company to divulge, disclose or make accessible such information. Nothing in this Agreement is intended to interfere a Party's right to (i) report possible violations of state or federal law or regulation to any governmental agency or entity, (ii) make other disclosures that are protected under the whistleblower provisions of state or federal law or regulation, (iii) file a claim or charge with the U.S. Equal Employment Opportunity Commission ("*EEOC*"), any state human rights commission, or any other government agency or entity, or (iv) testify, assist, or participate in an investigation, hearing, or proceeding conducted by the EEOC, any state human rights commission, or any other government or law enforcement agency, entity or court. However, Executive understands and agrees that if Executive, or someone acting on Executive's behalf, should file, or cause to be filed, any claim, charge, complaint, or action against the Company and/or any other relief from the Company and/or any other Company Released Parties subject to the release of claims in this <u>Section 1</u>. Executive expressly waives any and all rights to recover any damages or other relief from the Company and/or any other government agency or entity or here therefor by accessing the relief to the secutive hereby waives the right to recover any damages or benefits in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other government agency or entity or in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other

(iii) Executive is hereby notified in accordance with the Defend Trade Secrets Act of 2016 that Executive will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that is made in confidence to a federal, state, or local government official, either directly or indirectly, or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, or is made in a complaint or other document that is filed under seal in a lawsuit or other proceeding. If Executive files a lawsuit for retaliation against the Company for reporting a suspected violation of law, Executive may disclose the Company's trade secrets to Executive's attorney and use the trade secret information in the court proceeding if Executive files any document containing the trade secret under seal, and does not disclose the trade secret, except pursuant to court order;

(iv) This Release does not release any claim that may not be waived by applicable law; and

(v) This Release does not release any claim relating to the Parties' obligations under the Agreement or this Release, the Parties' rights to enforce any of the Surviving Covenants; and any claims arising after the Release Effective Date.

2. <u>Knowing and Voluntary Waiver of Age Claims; Time for Consideration</u>. Executive, by Executive's free and voluntary act of signing below, acknowledges that (a) Executive has been given a period of 21 days (*"Release Review Period"*) to consider whether to agree to the terms contained in this Release, (b) the Company is advising and has advised Executive in writing (i.e., through this Release) to

RELEASE OF CLAIMS

Page 3 of 6

consult with an attorney of Executive's own choosing at Executive's cost, regarding the effect of this Release, and Executive has had a reasonable opportunity to do so, if so desired, (c) Executive understands that this Release specifically releases and waives all rights and claims Executive may have under the Age Discrimination in Employment Act ("*ADEA*") prior to the date on which Executive signs this Release, and (d) Executive knowingly and voluntarily agrees to all of the terms of this Release and intends to be legally bound thereby. The Parties agree that any changes to this Release, whether material or immaterial, will not restart the running of the Release Review Period.

Executive may not execute this Release before the Separation Date. If Executive does not sign this Release on (but not before) the Separation Date (or if later, before the Release Review Period expires), this Release offer will be withdrawn automatically. Executive understands that this Release Review Period may be waived and Executive can execute this Release on the Separation Date (but not before) even if the Release Review Period has not yet expired. If Executive executes this Release on the Separation Date and it is prior to expiration of the Release Review Period, Executive agrees that the decision to do so was voluntary and that he took the reasonable time necessary to consult with counsel. The procedure for Executive to accept this Release is to return a fully executed, dated, and witnessed Agreement to Michael Silver, Vice President and General Counsel, prior to the last day of the Release Review Period.

Executive further understands that Executive may revoke this Release within seven days after signing it, in which case this Release and all obligations in the Agreement are null and void. Revocation is only effective if Executive delivers a written notice of revocation to the Company, ATTN: Michael Silver, Vice President and General Counsel, within seven days after executing the Release. Executive understands that the Company's obligations under this Release do not become effective until after the seven-day revocation period has expired. This Release will become effective, enforceable and irrevocable on the eighth day after the date on which it is executed by Executive (the "*Release Effective Date*").

3. <u>Known Violations</u>. Executive represents and warrants that Executive is not aware of any illegal acts committed by or on behalf of the Company and represents that if Executive is or had been aware of any such conduct, that Executive has properly reported the same to the Board or the Company's Vice President and General Counsel in writing. Executive further represents and warrants that Executive is not aware of any (i) violations, allegations or claims that the Company has violated any federal, state, local or foreign law or regulation of any kind, or (ii) any facts, basis or circumstances relating to any alleged violations, allegations or claims that the Company has violated any federal, state, local or foreign law or regulation of any kind. If Executive learns of any such information, Executive shall immediately inform the Company's General Counsel.

4. Entire Agreement. This Release and the Agreement set forth the entire agreement between the Parties concerning the subject matter in this Release, and fully supersede any and all prior and contemporaneous promises, negotiations, understandings, representations, writings, discussions and/or agreements between the Parties, whether oral or written, pertaining to the subject matter of this Release and Executive's employment with the Company, apart from the Agreement. No oral statements or other prior written material not specifically incorporated into this Release shall be of any force or effect, and no changes in or additions to this Release shall be recognized, unless incorporated into this Release by written amendment, such amendment to become effective on the date stipulated in it. Any amendment to this Release must be signed by all Parties to this Release.

5. <u>Disclaimer of Reliance</u>. Executive acknowledges that in executing this Release, (a) Executive does not rely, and has not relied, upon any oral or written representation, promise or inducement by the Company and/or any of the other Company Released Parties, except as expressly contained in this Release, (b) Executive has relied solely on his own judgment (and if applicable, that of his own counsel),

RELEASE OF CLAIMS

Page 4 of 6

in entering into this Release, and (c) there is no presumption regarding the interpretation or construction of this Release against its drafter. Executive understands and agrees that Executive is precluded from bringing any fraud or similar claim against the Company or any of the other Company Released Parties associated with any such communications, representations, promises or inducements.

6. <u>Choice of Law and Forum</u>. This Release shall be governed by, and construed and interpreted in accordance with, the laws of the State of Texas, but without regard to principles of conflict of laws that might direct the application of the law of another forum. Any claim or dispute between the Parties shall be governed by the arbitration provision in the General Provisions of the Employment Agreement. To the extent not covered by such arbitration provision, any action to enforce the provisions of the Agreement or this Release, or otherwise relating to the Agreement or this Release, must be brought in any court of competent jurisdiction in Harris County, Texas, and the Parties hereby waive any objection to such venue including, without limitation, that it is inconvenient for either Party.

7. <u>Waiver of Jury Trial</u>. THE PARTIES WAIVE ANY RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS RELEASE OR ANY CLAIM HEREUNDER, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES TO IRREVOCABLY WAIVE TRIAL BY JURY, AND THAT ANY PROCEEDING WHATSOEVER BETWEEN THEM RELATING TO THIS RELEASE OR ANY CLAIM HEREUNDER SHALL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.

8. <u>Severability</u>. The Company and Executive agree that should any government agency or court of competent jurisdiction declare or determine that any provision of this Release is illegal or invalid, the validity of the remaining parts, terms or provisions of this Release shall not be affected and such provisions shall remain in full force and effect and any illegal or invalid part, term, or provision, will not be deemed to be a part of this Release.

9. <u>Counterparts</u>. The Company and Executive agree that this Release may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall be deemed one and the same instrument.

[Signature Page Follows]

RELEASE OF CLAIMS

Page 5 of 6

Please review this document carefully as it includes a release of claims.

IN WITNESS WHEREOF, Executive has entered into this Release, and the Company has caused this Release to be executed in its name and on its behalf by its duly authorized officer.

Although Executive may sign this Release prior to the end of the 21-day Release Review Period, Executive may not sign this Release on or before Executive's Separation Date.

EXECUTIVE WITNESS **Cary Bounds** Witness Signature Date: Title: Printed Name: ------Date: VAALCO ENERGY, INC. By:___ Printed Name: Title: Date: Address for notices: VAALCO Energy, Inc.

9800 Richmond Ave. Suite 700 Houston, TX 77042 Attn: Michael Silver Vice President and General Counsel

RELEASE OF CLAIMS

Page 6 of 6

EXECUTIVE EMPLOYMENT AGREEMENT

BETWEEN

VAALCO ENERGY, INC.

AND

GEORGE MAXWELL

(EFFECTIVE AS OF APRIL 19, 2021)

1

1

1

1

2

3

3

3

4

4

4

4

4

5 5

5

5

5 5

6 6

6

7

7 7

8

8

9

10

10

11

11

12

12

12

13

14

Article 1. EMPLOYMENT AND DUTIES

- 1.1 Definitions
- 1.2 Employment; Effective Date
- 1.3 Positions
- 1.4 Duties and Services
- 1.5 Other Interests
- Duty of Loyalty 1.6
- 1.7 **Executive Warranties**

Article 2. TERM AND TERMINATION OF EMPLOYMENT

- 2.1 Term of Employment
- 2.2 Notice of Termination
- 2.3 Resignation of Offices
- 2.4 Summary Termination (for Cause)

Article 3. COMPENSATION AND BENEFITS

- Base Salary 3.1
- Annual Bonuses 3.2
- 3.3 Equity Awards after the Effective Date
- 3.4 Business and Entertainment Expenses
- 3.5 Vacation
- 3.6 Health Benefits
- 3.7 Pension Benefits
- 3.8 Incapacity
- 3.9 Other Paid Leave
- 3.10 Training
- 3.11 Disciplinary & Grievance Procedures

Article 4. RIGHTS AND PAYMENTS UPON TERMINATION

- 4.1 Rights and Payments upon Termination 4.2 Limitation on Other Severance Benefits 4.3 Settlement Agreement 4.4 Notice of Termination 4.5 Payment in Lieu of Notice 4.6 Garden Leave Article 5. CONFIDENTIAL INFORMATION AND RESTRICTIVE COVENANTS 5.1 Access to Confidential Information and Specialized Training
 - 5.2 Agreement Not to Use or Disclose Confidential Information 5.3 Duty to Return Company Documents and Property 5.4 Intellectual Property
 - i

5.5	Inventions	14
5.6	Post Termination Restrictions	15
5.7	Forfeiture of Severance Payment	17
5.8	Removal of Restrictions by Court	18
5.9	No Previous Restrictive Agreements	18
5.10	0 Conflicts of Interest	18
5.11		18
5.12		19
5.13		19
5.14		19
5.15	5 Collective Agreements	19
Article 6	6. GENERAL PROVISIONS	20
6.1	Matters Relating to Section 409A of the Code	20
6.2	Withholdings; Right of Deduction	20
6.3	Nonalienation	20
6.4	Incompetent or Minor Payees	20
6.5	Indemnification	20
6.6	Successors and Assigns	21
6.7	Notice	21
6.8		22
6.9		22
6.10	0 Waiver of Breach	22
6.11		22
6.12		22
6.13	1	22
6.14	8	23
6.15		23
6.16	6 Counterparts	23
Appendi	ix A	A-1
Appendi		B-1
-pp-nu		DI

ii

EXECUTIVE EMPLOYMENT AGREEMENT

THIS EXECUTIVE EMPLOYMENT AGREEMENT (the "Agreement"), shall be effective as of April 19, 2021 (the 'Effective Date") and is made and entered into by and between VAALCO Energy, Inc., a Delaware corporation (hereafter "Company") and George Maxwell (hereafter "Executive"). The Company and Executive may sometimes hereafter be referred to singularly as a "Party" or collectively as the "Parties." The addresses of the Parties are set forth in the Notice section of this Agreement.

WITNESSETH:

WHEREAS, the Company desires to secure the employment services of Executive subject to the terms and conditions hereafter set forth; and

WHEREAS, Executive is willing to enter into this Agreement upon the terms and conditions hereafter set forth;

NOW, THEREFORE, in consideration of Executive's employment with the Company, and the mutual promises, covenants and obligations contained herein, the Parties hereby agree as follows:

Article 1. EMPLOYMENT AND DUTIES

1.1 <u>Definitions</u>. In addition to the terms defined in the text hereof, terms with initial capital letters as used herein have the meanings assigned to them, for all purposes of this Agreement, in the Definitions Appendix hereto, unless the context reasonably requires a broader, narrower or different meaning. The Definitions Appendix, as attached hereto, is part of this Agreement and incorporated herein.

1.2 Employment; Effective Date. The Executive's employment under this Agreement shall commence as of the Effective Date and continue for the Employment Period (as defined in <u>Section 2.1</u>). Executive has no previous period of continuous employment with Company. The Executive's employment by the Company shall be subject to the terms and conditions of this Agreement. No probationary period applies to the employment of Executive.

1.3 <u>Positions</u>. As of the Effective Date, the Executive will serve as the Chief Executive Officerof the Company ("**CEO**") and, in due course, as required by the Company, CEO of any such directly or indirectly fully-owned subsidiary of the Company that is domiciled in the United Kingdom (and to which this agreement may be assigned or transferred pursuant to Section 6.6 below) and Executive hereby consents to such assignment or transfer of his employment. In addition, Executive shall serve as a member of the Board of Directors of the Company (the "Board of Directors"). Subject to the terms and conditions of this Agreement, Executive shall serve in the position of CEO and/or in such other position or positions as the Parties may mutually agree, during the Employment Period. In addition, the Company shall nominate the Executive for re-election to the Board of Directors as and when his term expires during the Employment Period,

unless otherwise determined by the Board of Directors. If the Executive is not nominated for re-election then the last day of his membership of the Board shall be his Termination Date, and the Executive shall be entitled to the rights and payments pursuant to <u>Article 4</u> to which he is then entitled subject to any conditions applying to such payments, and in particular if the failure to nominate the Executive for re-election was due to Cause (as defined in <u>Section 2.4</u>) then the Executive will only be entitled to the payments in<u>Section 2.4</u>

1.4 Duties and Services. The Executive agrees to serve in the positions referred to inSection 1.3 and to perform diligently and faithfully to the best of his abilities the duties and services appertaining to such offices, as well as such additional duties and services appropriate to such offices upon which the Parties mutually may agree from time to time or that are assigned to him by the Board of Directors. The Executive's employment shall also be subject to the policies maintained and established by the Company from time to time, as the same may be amended or otherwise modified.

Executive shall at all times use his best efforts to in good faith comply with English, United States and foreign laws applicable to Executive's actions on behalf of the Company and its Affiliates. Executive's normal place of work is in such place in London, UK as may be specified as Company offices (from time to time) or such other place where the Board of Directors may reasonably require for the proper exercise and performance of his duties. Executive understands and agrees that he may be required to travel extensively at times for purposes of the Company's business including, but not limited to, travel to the Company's corporate offices in the Houston, Texas, metropolitan area, from time to time, upon request by the Board of Directors or according to the needs to the business. This may include working outside of the UK for a period or periods in excess of one month.

Executive's normal working hours shall be from 9:00 a.m. to 6:00 p.m. GMT on Mondays through Fridays. These hours and days are not variable, although Executive may be required to work such additional hours as are necessary for the proper performance of his duties. Executive acknowledges that he shall not receive additional remuneration in respect of such additional hours. The Parties each agree that the nature of Executive's position is such that his working time cannot be measured and, accordingly, that his employment falls within the scope of regulation 20 of the UK Working Time Regulations 1998 (*SI 1998/1833*).

The Company takes a zero-tolerance approach to tax evasion. Executive must not engage in any form of facilitating tax evasion, whether under UK law or under the law of any foreign country. Executive must immediately report to the Board of Directors any request or demand from a third party to facilitate the evasion of tax or any concerns that such a request or demand may have been made. Executive must at all times comply with the Company's "Anti-Bribery and Anti-Tax Avoidance Compliance Policy", a copy of which is available from the Company's General Counsel (the 'General Counsel') and which may be updated from time to time.

The Company confirms that the Executive will not be required to invest more than three times his Base Salary in the stock of the Company, over the five-year period from the Effective Date, irrespective of any other policy or practice in effect at the Company from time to time.

1.5 Other Interests. The Executive agrees that, during the Employment Period, unless prevented by Incapacity, he will devote his whole time and attention, energy and best efforts to the business and affairs of the Company and its Affiliates, and not to engage, directly or indirectly, in any other business or businesses, whether or not similar to that of the Company or an Affiliate, except with the consent of the Board of Directors. The foregoing notwithstanding, the Parties recognize and agree that the Executive may engage in passive personal investments (such as real estate investments and rental properties) and other civic and charitable activities (such as continued service on non-profit and/or educational boards) that do not conflict with the business and affairs of the Company or interfere with the Executive's performance of his duties hereunder without the necessity of obtaining the consent of the Board of Directors; provided, however, Executive agrees that if the Compensation Committee of the Board of Directors (the "Compensation Committee") determines that continued service with one or more civic or charitable entities is inconsistent with the Executive's duties hereunder and gives written notice to the Executive, he will promptly resign from such position(s). This Section 1.5 shall not apply to any interests with which the Executive is currently engaged and which have previously been disclosed to, and approved by, the Chairman of the Board of Directors, prior to the Effective Date. This Agreement shall not apply to restrict or curtail the Executive's duties for the Company of the Secutive's duties therewise such as the provided that, in respect of such interests, priority and precedence shall always be given to Executive's duties for the Company of the restrict or curtail the Executive's duties for the Company of the Board of Directors, prior to the Effective Date. This Agreement.

1.6 Duty of Loyalty. The Executive acknowledges and agrees that the Executive owes a fiduciary duty of loyalty, fidelity, and allegiance to use his best endeavours to act at all times in the best interests of the Company and its Affiliates. In keeping with these duties, the Executive shall make full disclosure to the Company of all business opportunities pertaining to the Company's business, and he shall not appropriate for the Executive's own benefit any business opportunity concerning the subject matter of such fiduciary relationship. Executive shall comply with any statutory, fiduciary or common law duties to the Company, of which he is a director and not take any action which would cause him to be disqualified from acting as a director. He shall use his best endeavours to promote, protect, develop and extend the Company business and comply with all reasonable and lawful directions given to him by the Board of Directors and promptly make such reports to the Board of Directors in connection with the Company's business or affairs on such matters or at such times as are reasonably required.

1.7 Executive Warranties. The Executive represents and warrants to the Company that, by entering into this agreement or performing any of his obligations under it, he will not be in breach of any court order or any express or implied terms of any contract or other obligation binding on him. The Executive warrants that he is entitled to work in the UK without any additional approvals and will notify the Company immediately if he ceases to be so entitled during the Employment Period. The Executive warrants that he is not subject to any restrictions which prevent him from holding office as a director.

Article 2. TERM AND TERMINATION OF EMPLOYMENT

2.1 Term of Employment. Unless sooner terminated pursuant to other provisions hereof, the Company agrees to employ the Executive for the period beginning on the Effective Date and ending on April 30, 2022 (the "Initial Term of Employment"). Beginning effective as of May 1, 2022 (the "Initial Extension Date"), the term of employment hereunder shall be extended automatically for an additional successiveone-year period as of such date and as of each annual anniversary of the Initial Extension Date that occurs while this Agreement remains in effect so that the remaining term is one year; provided, however, if, at any time prior to the date that is at least twenty-eight (28) days (subject to statutory minimum as per Notice of Termination in Section 2.2) before the Initial Extension Date or any annual anniversary thereof, either Party gives Notice of Termination to the other Party that no such automatic extension shall occur and the Executive's employment shall terminate on April 30 of the then-current one-year extension, then, for purposes of clarity, such termination shall not be a Severance Payment Event.

The Initial Term of Employment, and any extension of employment hereunder, shall be referred to herein as a **Term of Employment**." The entire period from the Effective Date through the date of Executive's termination of employment with the Company, for whatever reason, shall be referred to herein as the **"Employment Period**."

2.2 Notice of Termination. If the Company or the Executive desires to terminate the Executive's employment hereunder at any time as of, or prior to, expiration of the Term of Employment, with or without Cause, such Party shall do so by giving written "Notice of Termination" to the other Party, subject to the terms and conditions of this Agreement regarding the rights and duties of the Parties upon termination of employment, and provided, however, that no such action shall alter or amend any other provisions hereof or rights arising hereunder. No further renewals of the Term of Employment hereunder shall occur pursuant to Section 2.1 after the giving of such Notice of Termination.

Such Notice of Termination by either Party shall be provided with at least four weeks' notice. This minimum notice period shall increase by one week per year of employment after four years of continuous employment up to a maximum of 12 weeks' notice after 12 years of continuous employment.

2.3 <u>Resignation of Offices</u>. Notwithstanding any other provision of this Agreement, upon the termination of the Executive's employment hereunder for any reason, unless otherwise requested by the Compensation Committee, Executive shall immediately resign from the Board of Directors and from all officer positions and all boards of directors of any Affiliates of which he may be a member. The Executive hereby agrees to execute any and all documentation of such resignations upon request by the Company, but he shall be treated for all purposes as having so resigned his officeholdings upon termination of his employment, regardless of when or whether he executes any such documentation.

2.4 <u>Summary Termination (for Cause)</u>. The Company may terminate the employment of Executive with immediate effect without notice and with no liability to make any further payment to him (other than in respect of amounts accrued due at the date of termination) if Cause (as defined herein) has occurred or exists to terminate Executive and the Board of Directors have determined that to be the case.

The Company's rights under this Section are without prejudice to any other rights that the Company might have at law to terminate the employment of Executive or to accept any breach of this Agreement by him as having brought the Agreement to an end. Any delay by the Company in exercising its rights to terminate shall not constitute a waiver of these rights.

Article 3. COMPENSATION AND BENEFITS

3.1 <u>Base Salary</u>(a) . During the Employment Period, the Executive shall receive a minimum annual base salary of US\$450,000, which shall be prorated for any period of less than 12 months (the "**Base Salary**"). The Compensation Committee shall review the Executive's Base Salary on an annual basis and may, in its sole discretion, increase, but not decrease, the Base Salary, and references in this Agreement to "Base Salary" shall refer to annual Base Salary as so increased. The Base Salary shall accrue daily and be paid in equal installments in accordance with the Company's standard policy regarding payment of compensation to executives, but no less frequently than monthly and shall be subject to such deductions as the Company is required by law to make. The Executive will be paid through the Company's U.S. payroll in USD, unless and until the Parties agree otherwise.

3.2 <u>Annual Bonuses</u>. For the 2021 calendar year and subsequent calendar years during the Employment Period, the Executive shall be eligible to receive an annual cash bonus (the "**Annual Bonus**") under the Company's annual incentive cash bonus plan for executives or any successor incentive cash bonus plan (the "**Bonus Plan**"), in an amount to be determined by the Compensation Committee, based on performance goals established by the Compensation Committee, in its discretion, pursuant to the terms of the Bonus Plan, and with a target percentage (the "**Incentive Target Percentage**") of fifty percent (50%) of the Executive's annual Base Salary as in effect at the beginning of the calendar year and may scale up or down based on achievement of personal and corporate goals established by the Compensation Committee.

3.3 Equity Awards after the Effective Date. During the Employment Period on and after the Effective Date, the Executive shall be eligible for stock options or other incentive awards in accordance with normal competitive pay practices, on a basis no less favorable than the process and approach used for the Company's other senior executives, as determined by the Compensation Committee in its discretion. The annual long-term incentive award shall be up to fifty percent (50%) of Executive's Base Salary.

3.4 <u>Business and Entertainment Expenses</u>. Subject to the Company's standard policies and procedures with respect to expense reimbursement as applied to its executives generally, the Company shall reimburse the Executive for, or pay on behalf of the Executive, the reasonable and appropriate expenses incurred by the Executive for business related purposes, including dues and fees to industry and professional organizations and costs of entertainment and business development.

3.5 <u>Vacation</u>. The Company's "Holiday Year" is a calendar year. The Executive shall be entitled to 25 days of paid vacation, plus public and bank holidays in England and Wales, each Holiday Year in accordance with the Company's vacation policy, as in effect from time to time. Such vacation leave entitlement in the first and last years of Executive's employment shall be pro-rated accordingly. Executive shall not carry forward any accrued but untaken paid vacation entitlement to a subsequent Holiday Year unless he has been prevented from taking it in the

relevant Holiday Year by one of the following: a period of sickness absence or statutory maternity, paternity, adoption, shared parental, parental or parental bereavement leave. In cases of sickness absence, carry-over is limited to four weeks' vacation leave per Holiday Year less any leave taken during the Holiday Year that has just ended. Any such carried over vacation leave which is not taken within 6 months of the end of the relevant Holiday Year will be lost and forfeited. The Company shall not pay the Executive in lieu of untaken vacation leave except on termination of his employment. On termination of employment, the Company shall pay Executive in lieu of any accrued but untaken vacation leave for the Holiday Year in which termination takes place and any untaken days permitted to be carried forward from the preceding Holiday Year. The amount of the payment in lieu of accrued vacation time will be calculated as 1/260th of Base Salary for each untaken day of the entitlement. Executive Benefits.

3.6 <u>Health Benefits</u>. During each calendar year of the Employment Period, beginning on the Effective Date andpro-rated in the first and last calendar years of the Employment Period, as necessary, the Company shall pay to Executive an annual amount of US\$22,000, which amount shall be indexed annually, effective as of January 1 for each calendar year during the Employment Period, based on the UK Consumer Prices Index (CPI), as an allowance for him to enable his participation in health and welfare benefits, including the acquisition of private health insurance as he may elect in his discretion.

3.7 <u>Pension Benefits</u>. During each calendar year of the Employment Period, beginning on the Effective Date andpro-rated in the first and last calendar years of the Employment Period, as necessary, the Company shall pay to Executive an annual amount of US\$17,000, which amount shall be indexed annually, effective as of January 1 for each calendar year during the Employment Period, based on the UK Consumer Prices Index (CPI), as an allowance for him to invest in either a personal pension scheme or similar retirement funding as he may elect in his discretion.

3.8 <u>Incapacity</u>. If the Executive is absent from work due to Incapacity, he (or a person on his behalf) shall notify the Board of Directors of the reason for the absence as soon as possible but no later than 10:00 a.m. on the first day of absence. Subject to the Executive's compliance with this agreement and the Company's "Medical Sick Leave Policy" and "Extended Medical Leave Policy" Executive shall be entitled to receive his full Base Salary during any periods of sickness absence up to a maximum of two weeks in any 52-week period. This does not affect any entitlement Executive may have to receive Statutory Sick Pay ("SSP") for the same periods of sickness absence, although any sick pay Executive receives from the Company shall be inclusive of any SSP due to him. Executive's qualifying days for SSP purposes are Monday to Friday.

Executive agrees to consent to medical examinations (at the Company's expense) by a doctor nominated by the Company at any time should we so require. If the Incapacity is or appears to be occasioned by actionable negligence, nuisance or breach of any statutory duty on the part of a third party in respect of which damages are or may be recoverable, Executive shall immediately notify the Board of Directors of that fact and of any claim, settlement or judgment made or awarded in connection with it and all relevant particulars that the Board of Directors may reasonably require. Executive shall if required by the Board of Directors, co-operate in any related legal proceedings and refund to the Company that part of any damages or compensation recovered by Executive relating to the loss of earnings for the period of the Incapacity, as the Board of Directors may reasonably determine, less any costs borne by Executive in connection with the recovery of

such damages or compensation, provided that the amount to be refunded shall not exceed the total amount paid to Executive by the Company in respect of the period of Incapacity. The Company's rights to terminate this employment of Executive under the terms of this agreement apply even when such termination would or might cause Executive to forfeit any entitlement to sick pay or other benefits.

3.9 <u>Other Paid Leave</u>. Executive may be eligible to take the following types of paid leave, subject to any statutory eligibility requirements or conditions (and the Company's rules) applicable to each type of leave in force from time to time: statutory paternity leave; statutory adoption leave; shared parental leave; and parental bereavement leave. Further details of such leave and any pay during such leave are available from the General Counsel.

3.10 Training. Aside from the Specialized Training, no training will be provided to Executive during the Employment Period

3.11 <u>Disciplinary & Grievance Procedures</u>. Executive shall be subject to the Company's "Employee Handbook" and "Code of Business Conduct and Ethics", copies of which are available upon request by Executive from the General Counsel. Executive may also request a copy of the grievance and disciplinary procedures for Executive from the General Counsel. These rules and procedures do not form part of Executive's contract of employment.

If Executive wishes to appeal against a grievance decision he may do so in writing to the Board of Directors in accordance with Company procedure as set out in the grievance procedure for Executive. If Executive wishes to appeal against a disciplinary decision he may apply in writing to the Board of Directors in accordance with Company disciplinary procedure.

The Board of Directors may suspend Executive from any or all of his duties for no longer than is necessary to investigate any disciplinary matter involving him or so long as is otherwise reasonable while any disciplinary procedure against you is outstanding. During any period of suspension:

(a) Executive shall continue to receive his Base Salary and all contractual benefits in the usual way and subject to the terms of any benefit arrangement;

(b) Executive shall remain an employee and bound by the terms of this Agreement;

(c) Executive shall ensure that the Board of Directors knows where he will be and how he can be contacted during each working day (except during any periods taken as holiday in the usual way);

(d) the Board of Directors may exclude Executive from his place of work or any of the Company's or its Affiliate's other premises; and

(e) the Board of Directors may require Executive not to contact or deal with (or attempt to contact or deal with) any officer, employee, consultant, client, customer, supplier, agent, distributor, shareholder, adviser or other business contact of the Company or any Affiliate

Article 4. RIGHTS AND PAYMENTS UPON TERMINATION.

4.1 Rights and Payments upon Termination. Executive's right to compensation and benefits for periods after the date on which his employment terminates with the Company and all Affiliates (the "Termination Date") shall be determined in accordance with this <u>Article 4</u>, as follows:

(a) *Minimum Payments*. Executive shall be entitled to the following minimum payments under this <u>Section 4.1(a)</u>, in addition to any other payments or benefits to which he is entitled to receive under the terms of this Agreement or any employee benefit plan or program:

(i) his accrued and unpaid Base Salary through the Termination Date;

(ii) his accrued and unused vacation days through the Termination Date;

(iii) reimbursement of his reasonable business expenses that were incurred but unpaid as of the Termination Date; and

(iv) if the Executive is not required to work and has given or been given Notice of Termination, payment of Base Salary in lieu of notice pursuant to Sections 2.2 and 4.5.

Such salary, payment in lieu of notice (if applicable) and accrued vacation days shall be paid to Executive within five (5) Business Days following the Termination Date in a cash lump sum less applicable withholdings and any tax or National Insurance Contributions which the Company is required by law to deduct. Business expenses shall be reimbursed in accordance with the Company's normal policy and procedures.

(b) *Termination Benefits*. In the event that during the Term of Employment Executive incurs a Severance Payment Event, the following severance benefits shall be provided to Executive hereunder or, in the event of his death before receiving all such benefits, to his Designated Beneficiary following his death:

(i) <u>Additional Payment</u>. The Company shall pay to Executive as additional compensation (the "**Additional Payment**"), an amount equal to fifty percent (50%) (in the event of a Regular Severance Payment Event), or one hundred fifty percent (150%) (in the event of a CIC Severance Payment Event), multiplied by the aggregate sum of the following compensation items:

(A) Executive's Base Salary as in effect as of the Termination Date; plus

(B) an amount equal to the greater of (i) the average of Executive's Annual Bonus (or other cash incentive bonus) paid or payable to Executive by the Company for the two calendar years immediately preceding the calendar year in which the Termination Date occurs; or (ii) Executive's target Annual Bonus for the full calendar year in which the Termination Date occurs; provided, however, in the event that the Termination Date occurs before the end of the calendar year, Executive shall be entitled to a prorata portion of the greater of clause (i) or (ii) above (based on the number of days in which he was employed during that year divided by 365).

Regardless of whether attributable to a Regular Severance Payment Event or a CIC Severance Payment Event, and subject to <u>Section 4.1(b)(ii)</u> in the event of an Anticipatory Termination, the Company shall make the Additional Payment to Executive over a six (6) month period in twelve (12) substantially equal bi-monthly payments that begin within twenty (20) days following the Termination Date. The payment of any Additional Payment shall be made in accordance with, and subject to, the Release requirements of <u>Section 4.3</u> and the Company's standard payroll procedures. In the event of Executive's death prior to the payment of all installments of the (1) Additional Payment as provided above, or (2) the Remaining Additional Payment Amount as provided in <u>Section 4.1(b)(ii)</u>, the remaining installment payments shall be aggregated and paid in a single sum payment to the Executive's Designated Beneficiary within sixty (60) days from Executive's date of death.

(ii) <u>Anticipatory Termination</u>. Notwithstanding any provision of this Agreement to the contrary, in the event of an Anticipatory Termination, the Company shall compute the Additional Payment payable to Executive as the result of a CIC Severance Payment Event and offset from such amount the aggregate amount of the installments of the Additional Payment, if any, that were already paid to Executive through the Change in Control Date as the result of his Regular Severance Payment Event. This Additional Payment attributable to the Executive's CIC Severance Payment Event, as offset by any installment payments already made to Executive in respect of a Regular Severance Payment Event through the Change in Control Date, is defined as the "**Remaining Additional Payment Amount**". The Remaining Additional Payment Amount shall be paid to Executive in substantially equal,bi-monthly installment payments over the remaining term of the one-year period that is specified in <u>Section 4.1(b)(i)</u>. The Remaining Additional Payment Amount shall be paid to Executive enter into a new Release Agreement.

4.2 Limitation on Other Severance Benefits.

(a) <u>Limitation on Other Severance Payments</u>. For purposes of clarity, in the event that (i) Executive voluntarily resigns or otherwise voluntarily terminates his own employment during the Term of Employment, *except for* (A) Good Reason or (B) due to his death or Disability, or (ii) Executive's employment is terminated due to a No Severance Benefits Event, then, in either such event under clause (i) or (ii), the Company shall have no obligation to provide the severance benefits described in <u>subsections (i) and (ii)</u> of <u>Section 4.1(b)</u>. Executive shall still be entitled to receive the severance benefits provided under <u>Section 4.1(a)</u>.

(b) <u>No Duplication of Severance Benefits</u> Notwithstanding <u>Section 4.1</u>, if Executive receives or is entitled to receive any severance benefit under any change of control policy, or any agreement with, or plan or policy of, the Company or any Affiliate, the amount payable under <u>Section 4.1(b)</u> to or on behalf of Executive shall be offset by such other severance benefits received by Executive, and Executive shall thus be entitled to receive the greater of such other severance benefits provided under this Agreement, and not any duplicate benefits. The severance payments provided under this Agreement shall also supersede and replace any duplicative severance benefits under any severance pay plan or program that the Company or any Affiliate maintains for employees generally and that otherwise may cover Executive.

4.3 Settlement Agreement. In order to receive the Termination Benefits, Executive must first execute a standard form Settlement Agreement in substantially the same form as the template document provided by the Company and which is appended to this Agreement as <u>Appendix B</u>. The Executive shall retain a Legal Adviser (as defined in the Settlement Agreement), as the Legal Adviser is also required to execute the Settlement Agreement. The Settlement Agreement will be modified as deemed to be necessary or appropriate to the circumstances of the Executive's termination and delivered to the Executive to sign a Settlement Agreement. Pursuant to the Settlement Agreement, Executive shall agree to release and waive, in return for such severance benefits, any employment-related claims or those claims related to his termination that he may have against the Company provided, however, the Settlement Agreement prior to the receipt thereof, and it shall not prevent the Executive enforcing the terms of the Settlement Agreement once agreed, or from relying on the provisions of any post termination rights (including, but without limitation, <u>Section 6.5</u> below,) given to the Executive under this Agreement. The Executive must return the executed Settlement Agreement within fifteen (15) Business and returns the executed Settlement Agreement within the Executive, and his Legal Adviser (as defined in the Settlement Agreement), signs and returns the Company signs the Settlement Agreement within the Settlement Period, then the Executive shall be entitled to the Termination Benefits irrespective of whether the Company signs the Settlement Agreement.

If the Settlement Agreement delivery and non-revocation period spans two taxable years, the Termination Benefits will always be paid in the second taxable year. No Termination Benefits shall be payable or provided by the Company unless and until the Settlement Agreement has been fully executed by Executive and his Legal Adviser (as defined therein), if any, has not been revoked, and is no longer subject to revocation by Executive, and the Termination Benefits will be paid 30 days thereafter the last applicable event.

4.4 Notice of Termination. Any termination of employment by the Company or Executive shall be communicated by Notice of Termination, pursuant to <u>Section 2.2</u>, to the other Party.

4.5 <u>Payment in Lieu of Notice</u> Notwithstanding the other provisions in this <u>Section 4</u>, the Company may, in its sole and absolute discretion, terminate the Employment Period at any time and with immediate effect by notifying Executive that it is exercising its right under this section and that it will make, within 28 days, a payment in lieu of notice (the "**Payment in Lieu**") to Executive. The Payment in Lieu will be equal to Executive's Base Salary (as at the Termination Date) which he would have been entitled to receive under this agreement during the notice period referred to in Section 2.2</u> (or, if Notice of Termination has already been given, during the remainder of the notice period), less the usual income tax and National Insurance contributions. For the avoidance of doubt, the Payment in Lieu shall not include any element in relation to:

(a) any bonus or commission payments that might otherwise have been due during the period for which the Payment in Lieu is made;

(b) any payment in respect of benefits which Executive would have been entitled to receive during the period for which the Payment in Lieu is made; and

(c) any payment in respect of any holiday entitlement that would have accrued during the period for which the Payment in Lieu is made.

The Company may choose to pay any sums due under this <u>Section 4.5</u> in equal monthly installments until the date on which the notice period would have expired if notice had been given.

Executive shall have no right to receive a Payment in Lieu unless the Company has exercised its discretion in relation to this Section and nothing in this Section shall prevent the Company from terminating Executive's employment in breach. Notwithstanding this Section Executive shall not be entitled to any Payment in Lieu if the Company would otherwise have been entitled to terminate the Employment Period without notice on the basis of a For Cause termination. In that case the Company shall also be entitled to recover from Executive any Payment in Lieu (or any installments) already made.

4.6 <u>Garden Leave</u>. Following service of notice to terminate the Employment Period by either Party, or if Executive purports to terminate his employment under this Agreement in breach of contract, the Board of Directors may by written notice place Executive on Garden Leave for the whole or part of the remainder of the Executive's Notice Period. During any period of Garden Leave:

(a) Executive shall not be a director of the Company unless the Board of Directors determines otherwise;

(b) the Company shall be under no obligation to provide any work to Executive and may revoke any powers he holds on any Company or Affiliate's behalf;

(c) the Company may require Executive to carry out alternative duties or to only perform such specific duties as are expressly assigned to him, at such location (including home) as the Company may decide;

(d) Executive shall continue to receive his Base Salary and all contractual benefits in the usual way and subject to the terms of any benefit arrangement;

(e) Executive shall remain an employee and bound by the terms of this Agreement (including any implied duties of good faith and fidelity);

(f) Executive shall ensure that the Board of Directors knows where he will be and how he can be contacted during each working day (except during any periods taken as holiday in the usual way);

(g) the Company may exclude Executive from any of the Company's or Affiliate's premises; and

(h) the Company may require Executive not to contact or deal with (or attempt to contact or deal with) any officer, employee, consultant, client, customer, supplier, agent, distributor, shareholder, adviser or other business contact of the Company or any Affiliate.

Article 5. CONFIDENTIAL INFORMATION AND RESTRICTIVE COVENANTS

5.1 Access to Confidential Information and Specialized Training. In connection with his employment and continuing on an ongoing basis during the Employment Period, the Company and its Affiliates will give Executive access to Confidential Information, which Executive did not have access to or knowledge of before the execution of this Agreement. Executive acknowledges and agrees that all Confidential Information is confidential and a valuable, special and unique asset of the Company that gives the Company an advantage over its actual and potential, current and future competitors. Executive further acknowledges and agrees that Executive owes the Company a fiduciary duty to preserve and protect all Confidential Information from unauthorized disclosure or unauthorized use, that certain Confidential Information constitutes "trade secrets" under applicable laws, and that unauthorized disclosure or unauthorized use of the Confidential Information would irreparably injure the Company or any of its Affiliates.

The Company also agrees to provide Executive with Specialized Training, which Executive does not have access to or knowledge of before the execution of this Agreement and continuing on an ongoing basis during his employment.

5.2 Agreement Not to Use or Disclose Confidential Information Both during the term of Executive's employment and after his termination of employment for any reason (including wrongful termination), Executive shall hold all Confidential Information in strict confidence, and shall not use any Confidential Information except for the benefit of the Company or its Affiliates, in accordance with the duties assigned to Executive. Executive shall not, at any time (either during or after the term of Executive's employment), disclose any Confidential Information to any Person (except other Persons who have a need to know the information in connection with the performance of services for the Company or its Affiliate), or copy, reproduce, modify, decompile or reverse engineer any Confidential Information, or remove any Confidential Information from the Company's premises, without the prior written consent of the Compensation Committee, or permit any other Person to do so. Executive shall take reasonable precautions to protect the physical security of all documents and other material containing Confidential Information, whether now known or later to become known to Executive.

The Executive shall hold in a fiduciary capacity for the benefit of the Company all Confidential Information relating to the Company or any of its Affiliates, and their respective businesses, that has been obtained by the Executive during the Executive's employment by the Company and which is not public knowledge (other than by acts of the Executive or representatives of the Executive in violation of this Agreement).

Following the termination of the Executive's employment with the Company for any reason, the Executive shall not, without the prior written consent of the Company or as may otherwise be required by law or legal process, or in compliance with any legal regulatory or compliance obligation or process, communicate or divulge any such Confidential Information to any Person other than the Company and those designated by it. This Section shall not apply to any information which is already in, or comes into, the public domain other than through Executive's unauthorised disclosure; or any protected disclosure within the meaning of section 43A of the Employment Rights Act 1996.

The Company has and will disclose to the Executive, or place the Executive in a position to have access to or develop, trade secrets and Confidential Information of the Company or its Affiliates; and/or has and will place the Executive in a position to develop business goodwill on behalf of the Company or its Affiliates; and/or has and will entrust the Executive with business opportunities of the Company or its Affiliates. As part of the consideration for the compensation and benefits to be paid to the Executive hereunder; the Executive hereby agrees to accept these restrictions in this Section to protect the trade secrets and Confidential Information of the Company and its Affiliates that have been and will in the future be disclosed or entrusted to the Executive, the business goodwill of the Company and its Affiliates that has been and will in the future be disclosed or entrusted to the Executive; and as an additional incentive for the Company to enter into this Agreement, the Company and the Executive agree to the non-competition and the non-solicitation obligations set forth in this Agreement.

5.3 Duty to Return Company Documents and Property. Upon the termination of Executive's employment with the Company and its Affiliates, for whatever reason, or upon the request of the Board of Directors at any time, Executive shall immediately return and deliver to the Company any and all papers, books, records, documents, memoranda and manuals, e-mail, electronic or magnetic recordings or data, including all copies thereof, belonging to the Company or its Affiliate or relating to their businesses, in Executive's possession or under his control, and regardless of, whether prepared by Executive or others. If at any time after the Employment Period, Executive determines that he has any Confidential Information in his possession or under his control, Executive shall immediately return to the Company all such Confidential Information, including all copies (including electronic versions) and portions thereof. Executive, will, at the Company's request, provide without delay a written and signed confirmation of compliance with this Section.

Within one (1) day after the end of the Employment Period for any reason, the Executive shall return to Company all Confidential Information which is in his possession, custody or control.

5.4 <u>Intellectual Property</u>. Executive shall promptly disclose to the Company all ideas, inventions, computer programs, and discoveries, whether or not patentable or copyrightable, which he may conceive or make, alone or with others, during the Employment Period, whether or not during working hours, and which directly or indirectly:

(a) relate to matters within the scope, field, duties or responsibility of Executive's employment with the Company; or

(b) are based on any knowledge of the actual or anticipated business or interest of the Company or any of its Affiliates; or

(c) are aided by the use of time, materials, facilities or information of the Company or its Affiliate.

Executive shall assign to the Company, without further compensation, all rights, titles and interest in all such ideas, inventions, computer programs and discoveries in all countries of the world.

Executive recognizes that all ideas, inventions, computer programs and discoveries of the type described above, conceived or made by Executive alone or with others within six (6) months after termination of employment (voluntary or otherwise), are likely to have been conceived in significant part either while employed by the Company or as a direct result of knowledge Executive had of Confidential Information. Accordingly, Executive undertakes and agrees to inform the Company of any such ideas, inventions or discoveries without delay and the same shall be presumed to have been conceived during his employment with the Company, unless and until the contrary is clearly established by Executive may be required to provide written confirmation of the existence or absence (as applicable) of any such ideas, inventions or discoveries at any time up to six (6) months following the end of the Employment Period.

5.5 <u>Inventions</u>. Any and all writings, computer software, inventions, improvements, processes, procedures and/or techniques which Executive may make, conceive, discover, or develop, either solely or jointly with any other Person, at any time during the Employment Period, whether at the request or upon the suggestion of the Company or otherwise, which relate to or are useful in connection with any business now or hereafter carried on or contemplated by the Company or its Affiliate, including developments or expansions of its present fields of operations, shall be the sole and exclusive property of the Company or its Affiliate. Executive shall take all actions necessary so that the Company or its Affiliate can prepare and present applications for copyright or Letters Patent therefor and can secure such copyright or Letters Patent wherever possible, as well as reissue renewals, and extensions thereof, and can obtain the record title to such copyright or patents. Executive shall not be entitled to any additional or special compensation or reimbursement regarding any such writings, computer software, inventions, improvements, processes, procedures and techniques. Executive acknowledges that the Company or its Affiliate from time to time may have agreements with other Persons which impose obligations or restrictions on the Company or its Affiliate regarding inventions made during the course of work thereunder or regarding the confidential nature of such work. Executive agrees to be bound by all such obligations and restrictions and to take all reasonable action which is necessary to discharge the obligations of the Company or its Affiliate with respect thereto.

Executive hereby irrevocably waives all moral rights under the Copyright, Designs and Patents Act 1988 (and all similar rights in other jurisdictions) which he has or will have in any existing or future works referred to in <u>Sections 5.4</u> and <u>5.5</u>. Executive irrevocably appoints the Company to be his attorney in his name and on his behalf to execute documents, use his name and do all things which are necessary or desirable for the Company to obtain for itself or its nominee the full benefit of these <u>Sections 5.4</u> and <u>5.5</u>.

5.6 Post Termination Restrictions.

(a) In order to protect the Confidential Information and the business, client and trade connections and the stability of the workforce, to which the Executive has had access as a result of his employment, Executive covenants with the Company (on its own behalf and as trustee and agent for its Affiliates) that Executive shall not:

 (i) for 12 months after the Termination Date, howsoever caused, solicit or endeavour to entice away from the Company or any of its Affiliates the business or custom of a Restricted Customer with a view to providing goods or services to that Restricted Customer in competition with any Restricted Business;

(ii) for 12 months after the Termination Date, howsoever caused, in the course of any business concern which is or intends to be in competition with any Restricted Business, offer to employ or engage or otherwise endeavour to entice away from the Company or any of its Affiliates any Restricted Person;

(iii) for 12 months after the Termination Date howsoever caused, in the course of any business concern which is or intends to be in competition with any Restricted Business, employ or engage or otherwise facilitate the employment or engagement of any Restricted Person, whether or not such person would be in breach of contract as a result of such employment or engagement;

(iv) for 12 months after the Termination Date, howsoever caused, be involved in any Capacity with any business concern which is (or intends to be) in competition with the Restricted Business.

(v) for 12 months after the Termination Date, howsoever caused, be involved with the provision of goods or services to (or otherwise have any business dealings with) any Restricted Customer in the course of any business concern which is in competition with any Restricted Business; or

(vi) at any time after the Termination Date, represent himself as connected with the Company or any of its Affiliates in any Capacity, other than as a former employee, or use any registered business names or trading names associated with the Company or any of its Affiliates.

(b) None of the above restrictions in Section 5.6(a)(i) through (vi) shall prevent Executive from:

(i) holding an investment by way of shares or other securities of not more than five percent (5%) of the total issued share capital of any company, whether or not it is listed or dealt in on a recognised stock exchange; or

(ii) being engaged or concerned in any business concern insofar as his duties or work shall relate solely to geographical areasoutside of the Restricted Area; or

(iii) any business with which the Executive has been involved pursuant to Section 1.5.

(c) The restrictions imposed on Executive by this Section 5.6 apply to his acting:

(i) directly or indirectly; and

(ii) on his own behalf or on behalf of, or in conjunction with, any firm, company or person.

(d) The periods for which the restrictions above apply shall be reduced by any period that Executive spends on Garden Leave immediately before the Termination Date.

(e) If, during his employment under this Agreement or before the expiry of the last of the covenants in this section, Executive receives an approach or offer to be involved in any Capacity in a business which competes with any part or parts of the Company's business or its Affiliate's business with which he is or has been involved to a material extent during the Appointment, Executive shall:

(i) notify the Board of Directors in writing of the fact of the approach or offer and the identity of the person making the approach or offer as soon as possible; and

(ii) give the person making the offer a copy of this Section 5.6 within seven (7) days of the offer being made.

(f) The obligations contained in this Section are continuing obligations and shall also apply if, at any time subsequent to the relevant approach or offer being made, but before the expiry of the last of the covenants in this Section, the business making the offer or approach so competes with the Company's or its Affiliate's business.

(g) If, at any time during Executive's employment, one or more Restricted Persons have left their employment, appointment or engagement with the Company or its Affiliate to carry out services for a business concern which competes with, or is intended to compete with any Restricted Business, Executive will not at any time during the six months following the last date on which any of those Restricted Persons were employed or engaged by the Company or its Affiliate, be employed or engaged in any way with that business concern.

(h) The Parties entered into the restrictions in this Section having been separately legally advised.

(i) Each of the restrictions in this Section is intended to be separate and severable. If any of the restrictions shall be held to be void but would be valid if part of their wording were deleted, such restriction shall apply with such deletion as may be necessary to make it valid or effective.

(j) If the Executive's employment is transferred to any firm, company, person or entity (other than an Affiliate) (the **New Employer**") pursuant to the Transfer of Undertakings (Protection of Employment) Regulations 2006, Executive will, if required, enter into an agreement with the New Employer containing post-termination restrictions corresponding to those restrictions in this clause protecting the confidential information, clients and customer relationships, trade secrets and business connections of the New Employer.

(k) Executive will, at the Company's request and expense, enter into a separate agreement with any Affiliate in which he agrees to be bound by restrictions corresponding to those restrictions in this Section (or such of those restrictions as may be appropriate) in relation to that Affiliate.

5.7 Forfeiture of Severance Payment. A "Forfeiture Event" for purposes of this Agreement will occur if (a) Executive knowingly or inadvertently violates any of the covenants or restrictions contained in Sections 5.1 through 5.6, or (b) the Company learns of facts within one (1) year following Executive's Termination Date that, if such facts had been known by the Company as of the Termination Date, would have resulted in the termination of Executive's employment hereunder by the Company for Cause, or (c) the Executive fails or refuses to provide upon request the written confirmation sought by the Company in Section 5.4. In the event of a Forfeiture Event, and if within thirty (30) days of being notified by the Company in writing of the Forfeiture Event, Executive has not remedied, cured, corrected or mitigated the Forfeiture Event to the satisfaction of the Company (acting reasonably), then within a further 30 days thereafter of being notified by the Company of that fact, he shall repay to the Company would not the Additional Payment received by Executive pursuant to Section 5.7 is a material part of this Agreement without which the Company would not have entered into this Agreement. Executive further covenants and agrees that should all or any part or application of this Section 5.7 be held or found invalid or unenforceable for any reason whatsoever by a court of competent jurisdiction in an action between Executive and the Company, then Executive shall promptly pay to the Company the amount of the Additional Payment, or such lesser amount as shall be determined to be the maximum reasonable and enforceable amount by a court, as applicable.

5.8 <u>Removal of Restrictions by Court</u>. It is expressly understood and agreed that the Company and the Executive consider the restrictions contained in this <u>Article 5</u> to be reasonable and necessary to protect the Confidential Information, client and customer connections, trade, workforce stability and reasonable business interests of the Company or its Affiliates. Nevertheless, if any of the aforesaid restrictions are found by a court having jurisdiction to be unreasonable, or overly broad as to geographic area or time, or otherwise unenforceable, the Parties intend for the unenforceable restrictions therein set forth to be deleted by such court such that the Agreement shall be reasonable and enforceable and, as such, shall then be fully enforced in the geographic area and for the time period to the full extent permitted by law.

5.9 No Previous Restrictive Agreements. Executive represents that, except for agreements he disclosed in writing to the Company, he is not bound by the terms of any agreement with any previous employer or other Person to (a) refrain from using or disclosing any trade secret or confidential or proprietary information in the course of Executive's employment by the Company or (b) refrain from competing, directly or indirectly, with the business of such previous employer or any other Person. Executive further represents that his performance of all the terms of this Agreement and his work duties for the Company does not, and will not, breach any agreement to keep in confidence proprietary information, knowledge or data acquired by Executive in confidence or in trust prior to Executive's employment with the Company, and Executive will not disclose to the Company or induce the Company to use any confidential or proprietary information or material belonging to any previous employer or other Person.

5.10 <u>Conflicts of Interest</u>. In keeping with his fiduciary duties to Company, Executive hereby agrees that he shall not become involved in a conflict of interest, or upon discovery thereof, allow such a conflict to continue at any time during the Employment Period. Moreover, Executive agrees that he shall abide by the Company's Code of Conduct, as it may be amended from time to time, and immediately disclose to the Board of Directors any known facts which might involve a conflict of interest or potential conflict of interest arising of which the Board of Directors was not aware.

5.11 <u>Remedies</u>. Executive acknowledges that the restrictions contained in this <u>Article 5</u>, in view of the nature of the Company's business, are reasonable and necessary to protect the Company's legitimate business interests, and that any violation of this Agreement would result in irreparable injury to the Company. In the event of a breach or a threatened breach by Executive of any provision of <u>Article 5</u>, the Company shall be entitled to claim injunctive relief restraining Executive from the commission of any breach, and to recover the Company's attorneys' fees, costs and expenses related to the breach or threatened breach, including, without limitation, the recovery of money damages, attorneys' fees, and costs. These covenants and disclosures shall each be construed as independent of any other provisions in this Agreement, and the existence of any claim or cause of action by Executive against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of such covenants and agreements.

The Executive acknowledges that money damages would not be sufficient remedy for any breach of <u>Article 5</u> by the Executive, and the Company shall also be entitled to specific performance as an available remedy for any such breach or any threatened breach. The remedies provided in this <u>Section 5.11</u> shall not be deemed the exclusive remedies for a breach of <u>Article 5</u>, but shall be in addition to all remedies available at law or in equity.

5.12 No Disparaging Comments. Executive and the Company (and its Affiliates) shall each refrain from any criticisms or disparaging comments about each other or in any way relating to Executive's employment or separation from employment; provided, however, that nothing in this Agreement shall apply to or restrict in any way the communication of information by the Company or any of its Affiliates or by the Executive to any law enforcement agency, or any regulatory or compliance authority. The Company and Executive will not be in breach of this covenant solely by reason of testimony or disclosure that is required for compliance with applicable law or regulation or by compulsion of law. A violation or threatened violation of this prohibition may be enjoined by a court of competent jurisdiction. The rights under this provision are in addition to any and all rights and remedies otherwise afforded by law to the Parties.

Executive acknowledges that in executing this Agreement, he has knowingly, voluntarily, and intelligently waived any free speech, free association, free press or First Amendment to the United States Constitution (including, without limitation, any counterpart or similar provision or right under the Texas Constitution or any other state constitution which may be deemed to apply) rights to disclose, communicate, or publish disparaging information or comments concerning or related to the Company or its Affiliate; provided, however, nothing in this Agreement shall be deemed to prevent Executive from testifying fully and truthfully in response to a subpoena from any court or from responding to an investigative inquiry from any governmental agency.

5.13 <u>Company Documents and Property</u>. All writings, records, and other documents and things comprising, containing, describing, discussing, explaining, or evidencing any Confidential Information, and all equipment, components, parts, tools, and the like in Executive's custody, possession or control that have been obtained or prepared in the course of Executive's employment with the Company shall be the exclusive property of the Company, shall not be copied and/or removed from the premises of the Company, except in pursuit of the business of the Company, and shall be delivered to the Company, without Executive retaining any copies, extracts or parts or any electronic versions, immediately upon notification of the termination of Executive's employment or at any other time requested by the Company. The Company shall have the right to retain, access, and inspect all property of any kind in the office or premises of the Company.

5.14 <u>Data Protection</u>. The Company will collect and process information relating to Executive in accordance with the privacy notice which is attached to this agreement. Executive is required to sign and date the privacy notice and return to the General Counsel. Executive shall comply with the Company's data protection policy when handling personal data in the course of employment' including personal data relating to any employee, worker, contractor, customer, client, supplier or agent of the Company or its Affiliates. Executive will also comply with the Company's "Code of Business Conduct and Ethics" and "Computer, Network & Equipment Use Policy". Failure to comply with the Company's data protection policy or any of the policies listed above shall subject the Executive to discipline pursuant to the Company's disciplinary procedures and, in serious cases, may be treated as gross misconduct leading to summary dismissal.

5.15 <u>Collective Agreements</u>. There is no collective agreement which directly affects the Executive's employment.

Article 6. GENERAL PROVISIONS

6.1 <u>Matters Relating to Section 409A of the Code</u>. Notwithstanding any provision in this Agreement to the contrary, if the payment of any compensation or benefit provided hereunder (including, without limitation, any Termination Benefits) would be subject to additional taxes and interest under Section 409A of the Code ("Section 409A"), then the Company may, after consulting with Executive, reform such provision to comply with Section 409A or to preclude imposition of taxes or interest under Section 409A, to the full extent permitted under Section 409A:

6.2 Withholdings; Right of Deduction. The Company may withhold and deduct from any benefits and payments made or to be made pursuant to this Agreement at any time (a) all applicable taxes and national insurance contributions as may be required pursuant to any law or governmental regulation or ruling, (b) all other normal employee deductions made with respect to Company's employees generally, and (c) any advances made to Executive and owed to Company.

6.3 <u>Nonalienation</u>. The right to receive payments under this Agreement shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge or encumbrance by Executive, his dependents or beneficiaries, or to any other Person who is or may become entitled to receive such payments hereunder. The right to receive payments hereunder shall not be subject to or liable for the debts, contracts, liabilities, engagements or torts of any Person who is or may become entitled to receive such payments, nor may the same be subject to attachment or seizure by any creditor of such Person under any circumstances, and any such attempted attachment or seizure shall be void and of no force and effect.

6.4 <u>Incompetent or Minor Payees</u>. Should the Compensation Committee determine, in its discretion, that any Person to whom any payment is payable under this Agreement has been determined to be legally incompetent or is a minor, any payment due hereunder, notwithstanding any other provision of this Agreement to the contrary, may be made in any one or more of the following ways: (a) directly to such Person; (b) to the legal guardian or other duly appointed personal representative of the individual or the estate of such Person; or (c) to such adult or adults as have, in the good faith knowledge of the Compensation Committee, assumed custody and support of such Person; and any payment so made shall constitute full and complete discharge of any liability under this Agreement in respect to the amount paid.

6.5 <u>Indemnification</u>. The Company hereby agrees to indemnify Executive and hold Executive harmless to the fullest extent permitted under the organizational documents of the Company and applicable law against and in respect of any and all actions, suits, proceedings, claims, demands, judgments, costs, expenses (including reasonable attorney's fees), losses, and damages (including advancement of fees and expenses) resulting from Executive's good faith performance of Executive's duties and obligations with the Company hereunder. The Company shall accordingly cover Executive under directors' and officers' liability insurance both during and, while potential liability exists, after the Employment Period, in the same amount and to the same extent as the Company covers its other officers and directors. The foregoing obligations shall survive the termination of Executive's employment with the Company and shall be in addition to any other indemnification rights Executive is entitled to, under existing or future indemnification agreements or otherwise.

6.6 Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the Company and any successor of the Company (whether direct or indirect, by purchase, merger, consolidation or otherwise), and this Agreement shall inure to the benefit of and be enforceable by Executive's legal representatives. The Company shall require any successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of the Company to assume expressly and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform it if no such succession had taken place. As used in this Agreement, "Company" shall mean the Company as previously defined and any successor by operation of law or otherwise, as well as any successor to its business and/or assets as a foresaid which assumes and agrees to perform this Agreement. The Company has the right and authority, in its discretion, to assign this Agreement to a directly or indirectly fully-owned subsidiary of the Company that is domiciled in the United Kingdom and Executive hereby agrees to any such assignment or transfer. Except as provided in the preceding provisions of this <u>Section 6.6</u>, this Agreement, and the rights and obligations of the Parties hereunder, are personal in nature and neither this Agreement, nor any right, benefit, or obligation of either Party hereto, shall be subject to voluntary or involuntary assignment, alienation or transfer, whether by operation of law or otherwise, without the written consent of the other Party.

6.7 Notice. Each Notice or other communication required or permitted under this Agreement shall be in writing and transmitted, delivered, or sent by personal delivery, prepaid courier or messenger service (whether overnight or same-day), or prepaid certified and recorded mail (with return receipt requested), addressed (in any case) to the other Party at the address for that Party set forth below or under that Party's signature on this Agreement, or at such other address as the recipient has designated by Notice to the other Party.

To the Company:	VAALCO Energy, Inc.
	9800 Richmond Avenue, Suite 700
	Houston, Texas 77042
	Attention: General Counsel & Corporate Secretary
To Executive:	George Maxwell
	(as set forth below his signature)

Each Notice or communication so transmitted, delivered, or sent (a) in person, by courier or messenger service, or by certified and recorded mail (return receipt requested) shall be deemed given, received, and effective on the date delivered to or refused by the intended recipient (with the return receipt, or the equivalent record of the courier or messenger, being deemed conclusive evidence of delivery or refusal), or (b) by email or facsimile shall be deemed given, received, and effective on the date of actual receipt (with the confirmation of transmission being deemed conclusive evidence of receipt, except where the intended recipient has promptly Notified the other Party that the transmission is illegible). Nevertheless, if the date of delivery or transmission is not a Business Day, or if the delivery or transmission is after 4:00 p.m. (local time at the recipient) on a Business Day, the Notice or other communication shall be deemed given, received, and effective on the next Business Day.

6.8 <u>Severability</u>. It is the desire of the Parties hereto that this Agreement be enforced to the maximum extent permitted by law, and should any provision contained herein be held unenforceable by a court of competent jurisdiction, the Parties hereby agree and consent that such provision shall be modified or deleted to create a valid and enforceable provision to the maximum extent permitted by law; provided, however, if such provision cannot be reformed, it shall be deemed ineffective and deleted herefrom without affecting any other provision of this Agreement. This Agreement should be construed by limiting and reducing it only to the minimum extent necessary to be enforceable under then applicable law.

6.9 No Third Party Beneficiaries This Agreement shall be binding upon and inure to the benefit of the Parties hereto, and to their respective successors and permitted assigns hereunder, but otherwise this Agreement shall not be for the benefit of any Persons who are third parties.

6.10 <u>Waiver of Breach</u>. No waiver by either Party of a breach of any provision of this Agreement by the other Party, or of compliance with any condition or provision of this Agreement to be performed by the other Party, will operate or be construed as a waiver of any subsequent breach by the other Party or any similar or dissimilar provision or condition at the same or any subsequent time. The failure of either Party to take any action by reason of any breach will not deprive such Party of the right to take action at any time while such breach continues.

6.11 <u>Survival of Certain Provisions</u>. Wherever appropriate to the intention of the Parties, the respective rights and obligations of the Parties hereunder shall, as applicable, survive any termination or expiration of this Agreement or following the Executive's Termination Date.

6.12 <u>Entire Agreement; Amendment and Termination</u>. This Agreement contains the entire agreement of the Parties with respect to the matters covered herein; moreover, this Agreement supersedes all prior and contemporaneous agreements and understandings, oral or written, between the Parties concerning the subject matter hereof. This Agreement may be amended, waived or terminated only by a written instrument that is identified as an agreed amendment, waiver or termination hereto and that is executed by or on behalf of both Parties.

6.13 Interpretive Matters. In the interpretation of the Agreement, except where the context otherwise requires:

(a) **Headings.** The Agreement headings are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement.

(b) The terms "including" and "include" do not denote or imply any limitation.

(c) The conjunction "or" has the inclusive meaning "and/or".

(d) The singular includes the plural, and vice versa, and each gender includes each of the others.

(e) The term "month" refers to a calendar month.

(f) Reference to any statute, rule, or regulation includes any amendment thereto or any statute, rule, or regulation enacted or promulgated in replacement thereof.

(g) The words "herein", "hereof", "hereunder" and other compounds of the word "here" shall refer to the entire Agreement and not to any particular provision;

(h) All amounts referenced herein are in U.S. dollars.

6.14 Governing Law; Jurisdiction.

(a) All matters or issues relating to the interpretation, construction, validity, and enforcement of this Agreement shall be governed by the laws of the state of Texas, without giving effect to any choice-of-law principle that would cause the application of the laws of any jurisdiction other than Texas.

(b) Jurisdiction and venue of any action or proceeding relating to this Agreement or any Dispute shall be exclusively in the federal and state courts of competent jurisdiction in Houston or Harris County, Texas. By entering into this Agreement, Executive consents to be subject to personal jurisdiction in the federal courts and Texas state courts situated in Houston, Texas or Harris County, Texas.

6.15 <u>Executive Acknowledgment</u>. Executive acknowledges that (a) he is knowledgeable and sophisticated as to business matters, including the subject matter of this Agreement, (b) he has read this Agreement and understands its terms and conditions, (c) he has had ample opportunity to discuss this Agreement with his legal counsel prior to execution, and (d) no strict rules of construction shall apply for or against the drafter or any other Party. Executive represents that he is free to enter into this Agreement including, without limitation, that he is not subject to any covenant not to compete or other restrictive covenant that would conflict with his employment duties and covenants under this Agreement.</u>

6.16 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument. Each counterpart may consist of a copy hereof containing multiple signature pages, each signed by one Party hereto, but together signed by both Parties.

[Signature page follows.]



IN WITNESS WHEREOF, Executive has hereunto set his hand and Company has caused this Agreement to be executed in its name and on its behalf by its duly authorized officer, to be effective as of the Effective Date.

WITNESS:

Signature:	/s/ R. Buchan	Signature:	/s/ George Maxwell
Name:	R. Buchan	Name:	George Maxwell
Date:	9 April 2021	Date:	9 April 2021
Executive's Address for Notices:			

George Maxwell 10 Springdale Crescent, Bieldside, Aberdeen AB15 9FG

ATTEST:

By:	/s/ Michael G. Silver
Name:	Michael G. Silver
Title:	General Counsel
Date:	10 April 2021

EXECUTIVE:

Signature:	/s/ George Maxwell
Name:	George Maxwell
Date:	9 April 2021

COMPANY:

VAALCO ENERGY, INC.

By:	/s/ Andrew Fawthrop
Name:	Andrew Fawthrop
Title:	Chairman of the Board
Date:	9 April 2021

APPENDIX A

Definitions Appendix

1. "Affiliate" has the same meaning ascribed to such term in Rule 12b-2 under the Securities Exchange Act of 1934, as amended from time to time.

2. "Anticipatory Termination" means a Separation From Service of the Executive on or after August 1, 2021, and within the time period that begins on the first day of the month that is three (3) months immediately preceding the first day of the month containing the Change in Control Date and ends on the Change in Control Date but only if the Executive's Separation From Service was (a) due to a termination by the Company without Cause or (b) a termination by the Executive for Good Reason. For purposes of clarification and not limitation, a Separation From Service for Cause, or due to Executive's death or Disability or his voluntary resignation without Good Reason, is not an Anticipatory Termination.

3. "Arbitration Rules" means the Rules for Employment Arbitrations of the American Arbitration Association, as in effect at the time of arbitration of a Dispute.

4. "Board" means the then-current Board of Directors of the Company.

5. "Business Day" means any Monday through Friday, excluding any such day on which banks are authorized to be closed in Texas.

6. "Capacity" means as agent, consultant, director, employee, worker, owner, partner or shareholder.

7. "Cause" shall mean the termination by the Company of the Executive's employment with the Company in circumstances of:

(a) any act or omission of Executive which constitutes gross misconduct and which has been determined by the Board of Directors, acting reasonably, to be so;

(b) the conviction of the Executive of a criminal offence by a court of competent jurisdiction of a crime involving moral turpitude or which is a crime other than an offence under any road traffic legislation in the UK or elsewhere) for which a fine or non-custodial penalty is imposed;

(c) Executive being disqualified from acting as a director or resigning as a director from the Company without the prior written approval of the Board of Directors;

(d) the commission by the Executive of a material act of fraud or dishonesty upon the Company or any of its Affiliates, or any customer or supplier thereof;

(e) the misappropriation of any funds or property of the Company or any of its Affiliates, or any customer or supplier thereof, by the Executive;

(f) the Executive's gross negligence or gross incompetence or otherwise the willful and continued failure by the Executive to perform the material duties assigned to him that is not cured to the reasonable satisfaction of the Company within 30 days after written notice of such failure is provided to Executive by the Board or the Compensation Committee (or by an officer of the Company who has been designated by the Board or the Compensation Committee for such purpose);

(g) the engagement by the Executive in any direct and material conflict of interest with the Company or any of its Affiliates without compliance with the Company's conflict of interest policy (or the conflict of interest policy of any Affiliate of the Company, if any) then in effect;

(h) the violation by the Executive of the Company's "Code of Business Conduct and Ethics" or the Executive's serious breach of the Company's rules and policies (including but not limited to the "Employee Handbook", "Anti-Bribery and Anti-Tax Avoidance Compliance Policy", "Insider Trading Policy", "Computer, Network & Equipment Use Policy", and other policies and procedures related to Company rules on electronic communications);

(i) the engagement by the Executive, without the written approval of the Board or the Compensation Committee, in any material activity which competes with the business of the Company or any of its Affiliates or which would result in a material injury to the business, reputation or goodwill of the Company or any of its Affiliates;

(j) the Executive is declared bankrupt or makes any arrangement with or for the benefit of his creditors or have a county court administration order made against you under the County Court Act 1984; or

(k) the Executive is, in the opinion of a medical practitioner physically or mentally incapable of performing his duties and may remain so for more than three months and the medical practitioner has given a medical opinion to the Board of Directors to that effect; or

(1) ceases to be eligible to work in the UK.

8. "Change in Control" means the occurrence of any one or more of the following events:

(a) The acquisition by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act (a **Person**")) of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of fifty percent (50%) or more of either (i) the then outstanding shares of common stock of the Company (the "**Outstanding Company Stock**") or (ii) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of directors (the "**Outstanding Company Voting Securities**"); provided, however, that the following acquisitions shall not constitute a Change in Control: (i) any acquisition directly from the Company or any Subsidiary, (ii) any acquisition by the Company or any Subsidiary or by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Subsidiary, or (iii) any acquisition by any corporation pursuant to a reorganization, merger, consolidation or similar business combination involving the Company (a "**Merger**"), if, following such Merger, the conditions described in <u>Section 7.8(c)</u> (below) are satisfied;

(b) Individuals who, as of the Effective Date, constitute the Board of Directors of the Company (the **'Incumbent Board**'') cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a director subsequent to the Effective Date whose election, or nomination for election by the Company's shareholders, was approved by a vote of at least a majority of the directors then comprising the Incumbent Board shall be considered a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board;

(c) The consummation of a Merger involving the Company, unless immediately following such Merger, (i) substantially all of the holders of the Outstanding Company Voting Securities immediately prior to Merger beneficially own, directly or indirectly, more than fifty percent (50%) of the common stock of the corporation resulting from such Merger (or its parent corporation) in substantially the same proportions as their ownership of Outstanding Company Voting Securities immediately prior to such Merger and (ii) at least a majority of the members of the board of directors of the corporation resulting from such Merger (or its parent corporation) were members of the Incumbent Board at the time of the execution of the initial agreement providing for such Merger; or

(d) The sale consummation, or other disposition of all or substantially all of the assets of the Company, unless immediately following such sale or other disposition, (i) substantially all of the holders of the Outstanding Company Voting Securities immediately prior to the consummation of such sale or other disposition beneficially own, directly or indirectly, more than fifty percent (50%) of the common stock of the corporation acquiring such assets in substantially the same proportions as their ownership of Outstanding Company Voting Securities immediately prior to the consummation of such assets or disposition, and (ii) at least a majority of the members of the board of directors of such corporation (or its parent corporation) were members of the Incumbent Board at the time of execution of the initial agreement or action of the Board of Directors providing for such sale or other disposition of assets of the Company; or

(e) The approval by the stockholders of the Company or the Board of a plan for the complete liquidation or dissolution of the Company.

Notwithstanding the foregoing provisions of this Change in Control definition, to the extent that any payment (or acceleration of payment) under the Agreement is considered to be deferred compensation that is subject to, and not exempt under, Code Section 409A, then the term Change in Control hereunder shall be construed to have the meaning as set forth in Code Section 409A, but only with respect to the payment (or acceleration of payment) of such deferred compensation, and to the extent inconsistent with the foregoing provisions of this definition as determined by the Incumbent Board.

9. "Change in Control Date" means the first date upon which a Change in Control event occurs, provided that such date occurs (a) during either (i) the Employment Period or (ii) the three-month period following the end of the Employment Period as specified in the definition of "Anticipatory Termination" if applicable, and (b) is, in any event, on or after August 1, 2021.

10. "CIC Window Period" means (a) the time period beginning on the Change in Control Date and ending on the last day of the twelve (12) consecutive month period that begins immediately following the last day of the month containing the Change in Control Date, or (b) following an Anticipatory Termination, the occurrence of a Change in Control (which Change in Control must qualify as a "change in control event" within the meaning of Section 409A) within the three-month period that is specified in the definition of "Anticipatory Termination".

11. "Code" means the Internal Revenue Code of 1986, as amended, or its successor. References herein to any Section of the Code shall include any successor provisions of the Code.

12. "Confidential Information" means any information or material known to, or used by or for, the Company or an Affiliate (whether or not owned or developed by the Company or an Affiliate and whether or not developed by Executive) that is not generally known by other Persons in the Business. For all purposes of the Agreement, Confidential Information includes, but is not limited to, the following: all trade secrets of the Company or an Affiliate; all non-public information that the Company or an Affiliate has marked as confidential or has otherwise described to Executive (either in writing or orally) as confidential; all non-public information concerning the Company's or Affiliate's products, services, prospective products or services, research, prospects, leases, surveys, seismic data, drilling data, designs, prices, costs, marketing plans, marketing techniques, studies, test data, leasehold and royalty owners, investors, suppliers and contracts; all business records and plans; all personnel files; all financial information of or concerning the Company's operating system software, application software, software and system methodology, hardware platforms, technical information, inventions, computer programs and listings, source codes, object codes, copyrights and other intellectual property; all technical specifications; any proprietary information belonging to the Company or an Affiliate; all computer or software or software manuals of the Company or an Affiliate; all Company or Affiliate training or instruction manuals; all Company or Affiliate electronic data; and all computer system passwords and user codes.

13. "Designated Beneficiary" means Executive's surviving spouse, if any, as determined for purposes of the Code. If there is no such surviving spouse at the time of Executive's death, then the Designated Beneficiary shall be Executive's estate.

14. "Disability" shall mean that Executive is entitled to receive long-term disability ("LTD") income benefits under the LTD plan or policy maintained by the Company or an Affiliate that covers Executive. If, for any reason, Executive is not covered under such LTD plan or policy, then "Disability" shall mean a "permanent and total disability" as defined in Code Section 22(e)(3) and Treasury regulations thereunder. Evidence of such Disability shall be certified by a physician acceptable to both the Company and Executive. In the event that the Parties are not able to agree

on the choice of a physician, each shall select one physician who, in turn, shall select a third physician to render such certification. All costs relating to the determination of whether Executive has incurred a Disability shall be paid by the Company. Executive agrees to submit to any examinations that are reasonably required by the attending physician or other healthcare service providers to determine whether he has a Disability.

15. "Dispute" means any dispute, disagreement, controversy, claim, or cause of action arising in connection with or relating to this Agreement or Executive's employment or termination of employment hereunder, or the validity, interpretation, performance, breach, modification or termination of this Agreement.

16. "Good Reason" means, with respect to Executive, the occurrence of any one or more of the following events which first occurs during the Employment Period, except as a result of actions taken in connection with termination of Executive's employment for Cause or Disability, and without Executive's specific written consent:

(a) The assignment to Executive of any duties that are materially inconsistent with Executive's executive position, which in this definition includes status, reporting relationship to the Board of Directors, office, title, scope of responsibility over corporate level staff or operations functions, or responsibilities as an officer of the Company, or any other material diminution in Executive's position, authority, duties, or responsibilities, other than (in any case or circumstance) an isolated and inadvertent action not taken in bad faith that is remedied by the Company within thirty (30) Business Days after Notice thereof to the Company by Executive; or

(b) The Company requires Executive to be based at any office or location that is farther than forty (40) miles from Executive's principal office location located in the London, UK metropolitan area, except for required business travel; or

(c) Any failure by the Company to obtain an assumption of this Agreement by its successor in interest, or any action or inaction that constitutes a material breach by the Company of this Agreement.

Notwithstanding the foregoing definition of "Good Reason", Executive cannot terminate his employment under the Agreement for Good Reason unless Executive (1) first provides written Notice to the Compensation Committee of the event (or events) that Executive believes constitutes a Good Reason event (above) within sixty (60) days from the first occurrence date of such event, and (2) provides the Company with at least thirty (30) Business Days to cure, correct or mitigate the Good Reason event so that it either (A) does not constitute a Good Reason event hereunder or (B) Executive specifically agrees, in writing, that after any such modification or accommodation by the Company, such event does not constitute a Good Reason event hereunder.

17. "Garden Leave" means any period under which the Company has exercised its rights under Section 4.6.

18. "Incapacity" means any sickness, injury or other medical disorder or condition which prevents Executive from carrying out his duties.

19. "No Severance Benefits Event" means termination of Executive's employment under the Agreement for Cause.

20. "Notice" means a written communication complying with Section 6.7 ("Notify" has the correlative meaning).

21. "Notice of Termination" means a written Notice which (a) indicates the specific termination provision in the Agreement that is being relied upon, (b) to the extent applicable, sets forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of Executive's employment under the provision so indicated, and (c) if the Termination Date is other than the date of receipt of such Notice, specifies the termination date (which date shall be not more than sixty (60) days after the giving of such Notice). Any termination of Executive by the Company for Cause, or by Executive for Good Reason, shall be communicated by Notice of Termination to the other Party. The failure by Executive or the Company to set forth in the Notice of Termination any fact or circumstance that contributes to a showing of Good Reason or Cause shall not waive any right of such Party, or preclude such Party from asserting, such fact or circumstance in enforcing such Party's rights.

22. "Person" means any individual, firm, corporation, partnership, limited liability company, trust, or other entity, including any successor (by merger or otherwise) of such entity.

23. "Restricted Area" means any geographical area (a) in which the Company or any of its Affiliates has operations or otherwise undertakes its or their business or (b) in relation to which, as at the Termination Date, the Company or any of its Affiliates has taken any action that discloses an intent to have operations or undertake its or their business therein. For avoidance of doubt, the Restricted Area as of the Effective Date comprises Gabon and Equatorial Guinea including their respective territorial waters.

24. "Restricted Business" means those parts of the Company's or its Affiliate's business with which Executive was involved or concerned to a material extent in the 12 months before the Termination Date.

25. "Restricted Customer" means any firm, company, person or other entity who, during the twelve (12) months before the Termination Date, was a customer or prospective customer of or was in the habit of dealing with the Company or its Affiliate and with whom Executive had contact or about whom he became aware or informed or was introduced to by the Company or its Affiliate in the course of his employment. This definition shall not include any firm, person, company or other entity with whom the Executive had a prior relationship, or with whom he had worked, or done business with before the date of the commencement of this Agreement except with respect to any operations or other business of the Company or any of its Affiliates relating to the Restricted Area.

26. "Restricted Person" means anyone employed or engaged by the Company or its Affiliate at the level of manager, officer or above or who could materially damage the interest of the Company or its Affiliate if they were involved in any Capacity in any business concern which competes (or intends to compete) with any Restricted Business and with whom the Executive dealt in the twelve (12) months before the Termination Date in the course of his employment.

27. "Settlement Agreement" means a settlement agreement, in such form as is prepared and delivered by the Company to Executive. The Settlement Agreement shall not release any claim by or on behalf of Executive for any payment or other benefit which is required under this Agreement and not conditioned upon execution and non-revocation of the Settlement Agreement by Executive prior to the receipt thereof, except as may otherwise be agreed to by Executive.

28. "Severance Payment Event" means either a (a) "CIC Severance Payment Event" or (b) "Regular Severance Payment Event", as such terms are defined below.

(a) "CIC Severance Payment Event" means either: the Executive's Separation From Service with the Company and all Affiliates that occurs within the CIC Window Period, other than (1) voluntarily by the Executive unless such resignation is for Good Reason, (2) due to Executive's death or Disability, or (3) involuntarily by the Company for Cause. Notwithstanding any provision herein to the contrary, any Separation From Service of the Executive that (i) does not occur within the CIC Window Period or (ii) is otherwise not described in this <u>subsection (a)</u>, shall not be considered a CIC Severance Payment Event.

(b) **"Regular Severance Payment Event"** means a Separation From Service that is not a CIC Severance Payment Event and such Separation From Service is due to: (1) involuntarily termination of Executive's Employment by the Company, except due to a No Severance Benefits Event, (2) termination of Executive's Employment due to his death or Disability, or (3) termination of Executive's Employment for Good Reason.

For all purposes of this definition of "Severance Payment Event", any transfer of the Executive's Employment from the Company to an Affiliate, from an Affiliate to the Company, or from one Affiliate to another Affiliate, is not a Separation From Service of the Executive (though any such transfer might, depending on the circumstances, constitute or result in a Separation From Service by the Executive for Good Reason). Any termination by the Company of the Executive for Cause, or by the Executive for Good Reason, shall be communicated by Notice of Termination to the other Party.

29. "Specialized Training" includes the training the Company provides to Executive that is unique to its business and enhances Executive's ability to perform his job duties effectively, which includes, without limitation, orientation training, operation methods training, and computer and systems training.

30. "Subsidiary" means a corporation or other entity, whether incorporated or unincorporated, of which at least a majority of the voting securities is owned, directly or indirectly, by the Company.

31. "Termination Benefits" means the benefits described in Section 4.1(b).

32. "Termination Date" means the date on which Executive's employment terminates with the Company and all Affiliates.

APPENDIX B

Form Settlement Agreement

DATED

SETTLEMENT AGREEMENT

between

VAALCO ENERGY, INC.

and

GEORGE MAXWELL

CONTENTS

CLAUSE

1.	Interpretation	4
2.	Arrangements on termination	5
3.	Termination Benefits	6
4.	Legal fees	7
5.	Waiver of claims	7
6.	Executive indemnities	9
7.	Company property and information	10
8.	Executive warranties and acknowledgments	10
9.	Resignation from offices	11
10.	[Garden leave	11
11.	Restrictive Covenants	11
12.	Confidentiality and announcements	11
13.	Directors & Officers Insurance and Company Indemnity	13
14.	Entire agreement	14
15.	Variation	14
16.	Third party rights	14
17.	Governing law	14
18.	Jurisdiction	14
19.	Subject to contract and without prejudice	15
20.	Counterparts	15

SCHEDULE

Schedule 1	Calculation of the Termination Benefits	16
Schedule 2	Claims	17
Schedule 3	Advisers certificate	21
Schedule 4	Announcement	22
Schedule 5	Restrictive covenants	23

Parties

- (1) VAALCO ENERGY, INC., incorporated in the United States, State of Delaware ("Company / we / us")
- (2) George Maxwell of [INDIVIDUAL ADDRESS] ("Executive / you")

BACKGROUND

- (A) You [have been **OR** were] employed by us from [DATE], most recently as [POSITION] under an employment agreement (**'Employment** Agreement') dated [DATE].
- (B) Your employment with us [shall terminate **OR** terminated] on [DATE].
- (C) The parties have entered into this agreement to record and implement the terms on which they have agreed to settle any claims that are set out herein and which you have or may have in connection with your employment or its termination or otherwise against us or any Group Company (as defined below) or our or their officers, employees or workers, whether or not those claims are, or could be, in the contemplation of the parties at the time of signing this agreement, and including, in particular, the statutory complaints that you raise in this agreement.
- (D) The parties intend this agreement to be an effective waiver of any such claims and to satisfy the conditions relating to settlement agreements and compromise contracts in the relevant legislation.
- (E) We enter into this agreement for ourselves and as agent and trustee for all Group Companies and we are authorised to do so. It is the parties intention that each Group Company should be able to enforce any rights it has under this agreement, subject to and in accordance with the Contracts (Rights of Third Parties) Act 1999.

Agreed terms

1. Interpretation

The following definitions and rules of interpretation apply in this agreement.

1.1 Definitions:

Adviser: [NAME] of [FIRM].

Affiliate: has the same meaning ascribed to such term in Rule 12b-2 under the Securities Exchange Act of 1934, as amended from time to time

Board: the then-current board of directors of the Company (including any committee of the board duly appointed by it).

Confidential Information: has the meaning given in Section 12 of Appendix A to the Employment Agreement.

Group Company: the Company, its Affiliates, subsidiaries or holding companies from time to time and any subsidiary of any holding company from time to time.

Holding company: has the meaning given in clause 1.6.

Post-Employment Notice Pay: has the meaning given in section 402D of the Income Tax (Earnings and Pensions) Act 2003 (ITEPA).

Post-Employment Notice Period: has the meaning given in section 402E(5) of ITEPA.

Subsidiary: has the meaning in clause 1.6.

Termination Benefits: as defined in the Employment Agreement under Article 4.

- 1.2 The headings in this agreement are inserted for convenience only and shall not affect its construction.
- 1.3 A reference to a particular law is a reference to it as it is in force for the time being taking account of any amendment, extension, ore-enactment and includes any subordinate legislation for the time being in force made under it.
- 1.4 Unless the context otherwise requires, words in the singular shall include the plural and in the plural shall include the singular.
- 1.5 The Schedules shall form part of this agreement and shall have effect as if set out in full in the body of this agreement. Any reference to this agreement includes the Schedules.
- 1.6 A reference to a holding company or a subsidiary means a holding company or a subsidiary (as the case may be) as defined in section 1159 of the Companies Act 2006 [and a company shall be treated, for the purposes only of the membership requirement contained in sections 1159(1)(b) and (c), as a member of another company even if its shares in that other company are registered in the name of (a) another person (or its nominee), whether by way of security or in connection with the taking of security, or (b) as a nominee].

2. Arrangements on termination

2.1 Your employment with the Company [shall terminate OR terminated] on [DATE] ("Termination Date").

- 2.2 The parties acknowledge that pursuant to clause 4.3 of your Employment Agreement in order to receive the Termination Benefits (as defined in that Agreement) you are required to enter into this Settlement Agreement and confirm the waiver of the Claims (as defined below and contained herein accordingly). The parties agree that any other payments contractually due to you under your Employment Agreement, (whether pre or post termination), or those subsisting post termination contractual rights given to you by the Employment Agreement which are not expressly waived as set out herein, are unaffected by this Agreement. Your compliance with entry into and the terms of this Agreement means the Termination Benefits shall be paid to you hereunder, as stated and as previously as set out in the Employment Agreement.
- 2.3 At the Termination Date, you confirm that you [have OR will have] completed the period of notice to which you are entitled under Section 2.2 of the Employment Agreement and we confirm that we will [have paid OR will pay] your salary [and benefits] for that period. [The parties accordingly believe that your Post-Employment Notice Period and Post-Employment Notice Pay are nil.

3. Termination Benefits

- 3.1 Subject to and conditional on you complying with the terms of this agreement, including, without limitation, clause 12, we shall [within 30 days of the Termination Date or receipt by us of a copy of this agreement signed by you in the form as set out in Schedule 3, whichever is later], pay to you as consideration for your performance under this agreement and by way of compensation for the termination of your employment the Termination Benefits ("Termination Benefits") calculated in accordance with the terms of Article 4 of the Employment Agreement, as applicable and set out in Schedule 1. We will pay the Termination Benefits less all required deductions for tax and National Insurance contributions. In this regard, the parties believe the following to be correct:
 - (a) [No part of the Termination Benefits is taxable as Post-Employment Notice Pay.]**OR** [The [first £[AMOUNT] (or USD equivalent calculated in accordance with clause 3.4 below) of the] Termination Benefits is Post-Employment Notice Pay and is taxable as earnings. We shall accordingly deduct income tax and employee National Insurance contributions from it at the appropriate rate.]
 - (b) [The [[next £30,000 **OR** first £30,000 **OR** balance] (or USD equivalent calculated in accordance with clause 3.4 below) of the Termination Benefits will be tax free, as a termination award under the threshold within the meaning of sections 402A(1) and 403 of ITEPA.]
 - (c) [The balance of the Termination Benefits will be taxable as a termination award exceeding the threshold within the meaning of sections 402A(1) and 403 of ITEPA. We shall accordingly deduct income tax from it at the appropriate rate.]

- 3.2 You shall be responsible for any further tax and employees National Insurance contributions due in respect of the Termination Benefits, and you shall indemnify us in respect of such liability in accordance with clause 6.1.
- 3.3 [The Termination Benefits will be paid to you [before **OR** after] receipt by you of your P45.]
- 3.4 Where the Termination Benefits or any sums hereunder are indicated in GBP Sterling but are to be paid to you in US Dollars (although subject to UK taxation as General Earnings under ITEPA Sections 15 or 27) the amounts chargeable hereunder to tax shall be strictly the GBP Sterling equivalent at the date you are entitled to be paid those sums or the date they are paid (if earlier). The conversion between these currencies should be calculated using the daily spot exchange rate of the Bank of England in operation at the relevant date.

4. Legal fees

We shall pay the reasonable legal fees (up to a maximum of \pounds [AMOUNT] plus VAT) incurred by you in obtaining advice on the termination of your employment and the terms of this agreement, such fees to be payable to the Adviser on production of an invoice addressed to you but marked as payable by us.

5. Waiver of claims

- 5.1 You agree that the terms of this agreement are offered by us without any admission of liability on our part and are in full and final settlement of all and any claims or rights of action that you have or may have against us or any Group Company or our or its officers, employees or workers arising out of your employment with us or its termination, whether under common law, contract, statute or otherwise, whether such claims are, or could be, known to the parties or in their contemplation at the date of this agreement in any jurisdiction and including, but not limited to, the specific claims set out in Part A of Schedule 2 and also the claims specified in Part B of Schedule 2 (together "**the Claims**" and each of which claims in Part A and Part B is hereby waived by this clause).
- 5.2 The waiver in clause 5.1 shall not apply to the following:
 - (a) any claims by you to enforce this agreement; and
 - (b) claims in respect of personal injury of which you are not aware and could not reasonably be expected to be aware at the date of this agreement (other than claims under discrimination legislation);
 - (c) your existing entitlement to contractual payments or contractual rights given to you which subsist post-termination and in each case, which are set out in your Employment Agreement, pursuant to clause 2.2 above (and so are outside the ambit of this agreement);

(d) [if applicable, your claim to any Remaining Additional Payment Amount (as defined in the Employment Agreement under Section 4.1) if such entitlement arises after your entry into this Settlement Agreement].

5.3 You warrant that:

- (a) before entering into this agreement you received independent advice from the Adviser as to the terms and effect of this agreement and, in particular, on its effect on your ability to pursue the Claims specified in this agreement;
- (b) the Adviser has confirmed to you that they are a solicitor holding a current practising certificate and that there is in force a policy of insurance covering the risk of a claim by you in respect of any loss arising in consequence of their advice;
- (c) the Adviser shall sign and deliver to us a letter in the form attached as Schedule 3 to this agreement;
- (d) before receiving the advice you disclosed to the Adviser all facts and circumstances in respect of the Claims and/or that may give rise to a claim by you against us or any Group Company or our or its officers, employees or workers;
- (e) You have not filed (and will not file) any complaints, claims or actions against the Company or any Group Company or its or their officers, employees or workers with any court, agency or commission regarding matters encompassed by this agreement;
- (f) the only claims that you have or may have against the Company or any Group Company or our or its officers, employees or workers (whether at the time of entering into this agreement or in the future) relating to your employment with us or its termination are the Claims, and as specified in clause 5.1 and Schedule 2; and.
- (g) You are not aware of any facts or circumstances that may give rise to any claim against us or any Group Company or our or its officers, employees or workers other than the Claims and/or those claims specified in clause 5.1 and Schedule 2 which are hereby waived.

You acknowledge that we acted in reliance on these warranties when entering into this agreement. The Company acknowledges that the above warranties (f) and (g) shall not apply insofar as they relate to matters specifically excluded from Section 5.1 of this Settlement Agreement, pursuant to clause 5.2.

- 5.4 You acknowledge that the conditions relating to settlement agreements and compromise contracts under section 147(3) of the Equality Act 2010, section 77(4A) of the Sex Discrimination Act 1975 (in relation to claims under that Act and the Equal Pay Act 1970), section 72(4A) of the Race Relations Act 1976, paragraph 2 of Schedule 3A to the Disability Discrimination Act 1995, paragraph 2(2) of Schedule 4 to the Employment Equality (Sexual Orientation) Regulations 2003, paragraph 2(2) of Schedule 4 to the Employment Equality (Religion or Belief) Regulations 2003, paragraph 2(2) of Schedule 5 to the Employment Equality (Age) Regulations 2006, section 288(2B) of the Trade Union and Labour Relations (Consolidation) Act 1992, section 203(3) of the Employment Rights Act 1996, regulation 35(3) of the Working Time Regulations 1998, section 49(4) of the National Minimum Wage Act 1998, regulation 41(4) of the Transnational Information and Consultation of the Fixed-Term Employees (Prevention of Less Favourable Treatment) Regulations 2002, regulation 40(4) of the Information and Consultation of Employees Regulations 2002, paragraph 13 of the Schedule to the Occupational and Personal Pension Schemes (Consultation by Employees and Miscellaneous Amendment) Regulations 2006, regulation 62 of the Companies (Cross Border Mergers) Regulations 2007 and section 58 of the Pensions Act 2008 have been satisfied.
- 5.5 The waiver in clause 5.1 shall have effect irrespective of whether or not, at the date of this agreement, you are or could be aware of such claims or have such claims in your express contemplation (including such claims of which you become aware after the date of this agreement in whole or in part as a result of new legislation or the development of common law or equity).
- 5.6 You agree that, except for the Termination Benefits provided for in this agreement, and subject to the provisions in clause 5.2, and any continuing contractual entitlements under the Employment Agreement pursuant to clause 2.2, above you shall not be eligible for any further or different payment or compensation from us or any Group Company relating to your employment or its termination and you expressly waive any right or claim that you have or may have to such payment or arrangement or award you may have received had your employment not terminated.

6. Executive indemnities

6.1 You shall indemnify us on a continuing basis in respect of any income tax or National Insurance contributions (save for employers National Insurance contributions) due in respect of (a) the payments and benefits in clause 3.1 and (b) any Remaining Additional Payment Amount (if any) which may become payable to you (and any related interest, penalties, costs and expenses). We shall give you reasonable notice of any demand for tax which may lead to liabilities on you under this indemnity and shall provide you with reasonable access to any documentation you may reasonably require to dispute such a claim (provided that nothing in this clause shall prevent us from complying with our legal obligations with regard to HM Revenue and Customs or other competent body).

6.2 If you breach any provision of this agreement or pursue a claim against us or any Group Company arising out of your employment or its termination other than those excluded under clause 5, you agree to indemnify us or any Group Company for any losses suffered as a result thereof, including all reasonable legal and professional fees incurred.

7. Company property and information

- 7.1 Notwithstanding clause 14 and the termination of your employment, you acknowledge that:
 - (a) Sections 5.3, 5.4, 5.5, and 5.6 of the Employment Agreement will continue to apply after the Termination Date and you shall continue to comply with the same; and
 - (b) Sections 5.7 and 5.11 of the Employment Agreement shall continue to apply to any breach or threatened breach of Sections 5.3, 5.4, 5.5, and 5.6 of the Employment Agreement.
- 7.2 You shall, before the Termination Date, erase **OR** as at the date of this agreement, warrant that you have erased irretrievably any information relating to our or any Group Company's business or affairs or our or its business contacts from computer and communications systems and devices owned or used by you outside our premises, including such systems and data storage services provided by third parties (to the extent technically practicable).
- 7.3 You shall, if requested to do so by the Board of Directors, provide a signed statement that you have complied fully with your obligations under clause 7.1 and clause 7.2 and shall provide us with such reasonable evidence of compliance as may be requested.

8. Executive warranties and acknowledgments

- 8.1 As at the date of this agreement, you hereby warrant and represent to us that you have not undertaken any conduct which would be a Forfeiture Event as defined in Section 5.7 of the Employment Agreement and you are aware that any payment to you pursuant to clause 3 above is conditional on this being so.
- 8.2 You agree to make yourself available to, and to cooperate with, us or our advisers in any internal investigation or administrative, regulatory, judicial or quasi-judicial proceedings. You acknowledge that this could involve, but is not limited to, responding to or defending any regulatory or legal process, providing information in relation to any such process, preparing witness statements and giving evidence in person on our behalf. We shall reimburse any loss of earnings or reasonable expenses that you incur as a consequence of complying with your obligations under this clause, provided that such expenses are approved in advance by us.

9. Resignation from offices

- 9.1 You [acknowledge that you have resigned from your position as [POSITION] and] shall resign immediately from any [other] office, trusteeship or position that you hold in or on [our **OR** any Group Company's] behalf.
- 9.2 [You irrevocably appoint us to be your attorney in your name and on your behalf to sign, execute or do any such instrument or thing and generally to use your name in order to give us (or our nominee) the full benefit of the provisions of this clause.]

10. [Garden leave

10.1 During the period from [DATE] to the Termination Date you shall not perform any services for us or any Affiliate ('Garden Leave"), and shall comply with the provisions of Section 4.6 of the Employment Agreement accordingly.]

11. Restrictive covenants

- 11.1 Notwithstanding clause 14, you acknowledge that the post-termination restrictions in Article 5 of the Employment Agreement will continue to apply after the Termination Date [save that the period of each will be reduced by the period that you spend on Garden Leave] and which, for clarity are set out at Schedule 5 to this Agreement.
- 11.2 We shall pay £100 to you as consideration for your entering into the restrictive covenants in Schedule 5, such sum to be paid within 30 days after the Termination Date or receipt by us of a copy of this agreement signed by you and receipt by us of a letter from the Adviser as set out in Schedule 3, whichever is later. We shall deduct income tax and National Insurance contributions from this sum

12. Confidentiality and announcements

- 12.1 Notwithstanding clause 14, and the termination of your employment, you acknowledge that:
 - (a) Sections 5.1 and 5.2 of the Employment Agreement will continue to apply after the Termination Date and you shall continue to comply with the same; and
 - (b) Sections 5.7 and 5.11 of the Employment Agreement shall continue to apply to any breach or threatened breach of those Sections 5.1 and 5.2.

- 12.2 The parties confirm that they have kept and agree to keep the existence and terms of this agreement and the circumstances concerning the termination of your employment confidential, save only:
 - (a) as provided in clause 12.4, clause 12.5 and clause 12.6;
 - (b) [to the extent required to complete our investigation into your grievance raised [on [DATE]OR [in your letter dated [DATE]] and any subsequent disciplinary investigation or procedure]; or
 - (c) as required by law or regulatory practice.
- 12.3 The parties shall continue to comply with Section 5.12 of the Employment Agreement notwithstanding the termination of your employment. You acknowledge that Section 5.7 and 5.11 of the Employment Agreement shall continue to apply to any breach or threatened breach of Section 5.12. This clause is subject to clause 12.4, clause 12.5 and clause 12.6.
- 12.4 The parties are permitted to make a disclosure or comment that would otherwise be prohibited by clause 12.3 and clause 12.4 if, where necessary and appropriate:
 - (a) in your case, you make it to:
 - your spouse, civil partner or partner or immediate family provided that they agree to keep the information confidential; or
 - any person who owes you a duty of confidentiality (which you agree not to waive) in respect of information you disclose to them, including your legal or tax advisers or persons providing you with medical, therapeutic, counselling or support services; or
 - (iii) your insurer for the purposes of processing a claim for loss of employment; [or]
 - (iv) [your recruitment consultant or prospective employer to the extent necessary to discuss your employment history; [or]]
 - (v) [any government benefits agency for the purposes of you making a claim for benefits; and]
 - (b) in our case, we make it to:
 - (vi) our officers, employees or workers provided that they agree to keep the information confidential; or
 - (vii) any person who owes us a duty of confidentiality (which we agree not to waive) in respect of information we disclose to them, including, our legal, tax, compliance or other professional advisers.

- 12.5 Nothing in this clause 12 shall prevent you or any of our officers, employees, workers or agents from making a protected disclosure under section 43A of the Employment Rights Act 1996.
- 12.6 Nothing in this clause 12 shall prevent you or us (or any of our officers, employees, workers or agents) from:
 - (a) reporting a suspected criminal offence to the police or any law enforcement agency or co-operating with the police or any law enforcement agency regarding a criminal investigation or prosecution; or
 - (b) doing or saying anything that is required by HMRC or a regulator, ombudsman or supervisory authority; or
 - (c) whether required to or not, making a disclosure to, orco-operating with any investigation by, HMRC or a regulator, ombudsman or supervisory authority regarding any misconduct, wrongdoing or serious breach of regulatory requirements (including giving evidence at a hearing); or
 - (d) complying with an order from a court or tribunal to disclose or give evidence; or
 - (e) disclosing information to HMRC for the purposes of establishing and paying (or recouping) tax and national insurance liabilities arising from your employment or its termination; or
 - (f) making any other disclosure as required by law.
- 12.7 We will make an announcement on [the Termination Date**OR** signature of this agreement] in the form set out in Schedule 4 and neither party will make any statement to third parties (save as specified in clause 12.5, clause 12.6 or clause 12.7) which is inconsistent with that announcement.
- 12.8 We shall pay £100 to you as consideration for your entering into the restrictions in this clause 12, such sum to be paid within 30 days of the Termination Date or receipt by us of a copy of this agreement signed by you and receipt by us of a letter from the Adviser as set out in Schedule 3 whichever is later. We shall deduct income tax and National Insurance contributions from this sum.

13. Directors & Officers Insurance and Company Indemnity

13.1 Notwithstanding the termination of your employment, the Company shall continue to indemnify you subject to and in accordance with Section 6.5 of the Employment Agreement, as it does for its other directors and officers of similar standing, for acts undertaken during your employment.

14. Entire agreement

Each party on behalf of itself and, in our case, as agent for any Group Company acknowledges and agrees with the other party (with us acting on our own behalf and as agent for each Group Company) that:

- (a) This agreement and the Employment Agreement (where the latter is referenced in this agreement accordingly, and in particular by clause 2.2 above) together constitute the entire agreement between the parties and any Group Company and this agreement and the Employment Agreement (where the latter is referenced in this agreement) together supersede and extinguish all agreements, promises, assurances, warranties, representations and understandings between them whether written or oral, relating to the subject matter of this agreement;
- (b) in entering into this agreement, it does not rely on, and shall have no remedies in respect of, any statement, representation, assurance or warranty (whether made innocently or negligently) that is not set out in this agreement or referenced in this agreement; and
- (c) it shall have no claim for innocent or negligent misrepresentation or negligent misstatement based on any statement in this agreement.

15. Variation

No variation of this agreement shall be effective unless it is in writing and signed by the parties (or their authorised representatives).

16. Third party rights

Except as expressly provided elsewhere in this agreement, no person other than you and [us**OR** any Group Company] shall have any rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of this agreement. [This does not affect any right or remedy of a third party which exists, or is available, apart from that Act.]

17. Governing law

This agreement and any dispute or claim arising out of or in connection with it or its subject matter or formation (includingnon-contractual disputes or claims) shall be governed by and construed in accordance with the law of England and Wales.

18. Jurisdiction

Each party irrevocably agrees that the courts of England and Wales shall have [exclusive**OR** non-exclusive] jurisdiction to settle any dispute or claim arising out of or in connection with this agreement or its subject matter or formation (including non-contractual disputes or claims).

19. Subject to contract and without prejudice

This agreement shall be deemed to be without prejudice and subject to contract until such time as it is signed by both parties and dated, when it shall be treated as an open document evidencing a binding agreement.

20. Counterparts

This agreement may be executed in any number of counterparts, each of which shall constitute a duplicate original, but all the counterparts shall together constitute the one agreement.

This agreement has been entered into on the date stated at the beginning of it.

Schedule 2 Claims

Part A - Specific Claims

1. Claims:

[SET OUT SPECIFIC CLAIMS]

[Claims below to be divided into those **specific** claims which have been intimated by the Executive to his Legal Adviser (Part A) and included here and all other UK employment-related claims which are also being waived (Part B)].

Part B - Other Claims

- 2. Claims:
- 2.1 for breach of contract or wrongful dismissal;
- 2.2 for unfair dismissal, under section 111 of the Employment Rights Act 1996;
- 2.3 in relation to the right to a written statement of reasons for dismissal, under section 93 of the Employment Rights Act 1996;
- 2.4 for a statutory redundancy payment, under section 163 of the Employment Rights Act 1996;
- 2.5 in relation to an unlawful deduction from wages or unlawful payment, under section 23 of the Employment Rights Act 1996;
- 2.6 for unlawful detriment, under section 48 of the Employment Rights Act 1996 or section 56 of the Pensions Act 2008;
- 2.7 in relation to written employment particulars and itemised pay statements, under section 11 of the Employment Rights Act 1996;
- 2.8 in relation to guarantee payments, under section 34 of the Employment Rights Act 1996;
- 2.9 in relation to suspension from work, under section 70 of the Employment Rights Act 1996;
- 2.10 in relation to parental leave, under section 80 of the Employment Rights Act 1996;
- 2.11 in relation to a request for flexible working, under section 80H of the Employment Rights Act 1996;

- 2.12 in relation to time off work, under sections 51, 54, 57, 57B, 57ZC, 57ZF, 57ZH, 57ZM, 57ZQ, 60, 63 and 63C of the Employment Rights Act 1996;
- 2.13 in relation to working time or holiday pay, under regulation 30 of the Working Time Regulations 1998;
- 2.14 in relation to the national minimum wage, under sections 11, 18, 19D and 24 of the National Minimum Wage Act 1998;
- 2.15 for equal pay or equality of terms under sections 120 and 127 of the Equality Act 2010 and/or section 2 of the Equal Pay Act 1970;
- 2.16 for pregnancy or maternity discrimination, direct or indirect discrimination, harassment or victimisation related to sex, marital or civil partnership status, pregnancy or maternity or gender reassignment under section 120 of the Equality Act 2010 and/or direct or indirect discrimination, harassment or victimisation related to sex, marital or civil partnership status, gender reassignment, pregnancy or maternity under section 63 of the Sex Discrimination Act 1975;
- 2.17 for direct or indirect discrimination, harassment or victimisation related to race under section 120 of the Equality Act 2010 and/or direct or indirect discrimination, harassment or victimisation related to race, colour, nationality or ethnic or national origin, under section 54 of the Race Relations Act 1976
- 2.18 for direct or indirect discrimination, harassment or victimisation related to disability, discrimination arising from disability, or failure to make adjustments under section 120 of the Equality Act 2010 and/or direct discrimination, harassment or victimisation related to disability, disability-related discrimination or failure to make adjustments under section 17A of the Disability Discrimination Act 1995;
- 2.19 for direct or indirect discrimination, harassment or victimisation related to religion or belief under section 120 of the Equality Act 2010 and/or under regulation 28 of the Employment Equality (Religion or Belief) Regulations 2003;
- 2.20 for direct or indirect discrimination, harassment or victimisation related to sexual orientation, under section 120 of the Equality Act 2010 and/or under regulation 28 of the Employment Equality (Sexual Orientation) Regulations 2003;
- 2.21 for direct or indirect discrimination, harassment or victimisation related to age, under section 120 of the Equality Act 2010 and/or under regulation 36 of the Employment Equality (Age) Regulations 2006

- 2.22 for less favourable treatment on the grounds of part-time status, under regulation 8 of the Part-Time Workers (Prevention of Less Favourable Treatment) Regulations 2000;
- 2.23 for less favourable treatment on the grounds of fixed-term status, under regulation 7 of the Fixed-Term Employees (Prevention of Less Favourable Treatment) Regulations 2002;
- 2.24 under regulations 27 and 32 of the Transnational Information and Consultation of Employees Regulations 1999;
- 2.25 under regulations 29 and 33 of the Information and Consultation of Employees Regulations 2004;
- 2.26 under regulations 45 and 51 of the Companies (Cross-Border Mergers) Regulations 2007;
- 2.27 under paragraphs 4 and 8 of the Schedule to the Occupational and Personal Pension Schemes (Consultation by Employers and Miscellaneous Amendment) Regulations 2006;
- 2.28 under sections 68A, 87, 137, 145A, 145B, 146, 168, 168A, 169, 170, 174 and 192 of the Trade Union and Labour Relations (Consolidation) Act 1992;
- 2.29 in relation to the obligations to elect appropriate representatives or any entitlement to compensation, under the Transfer of Undertakings (Protection of Employment) Regulations 2006;
- 2.30 in relation to the right to be accompanied under section 11 of the Employment Relations Act 1999;
- 2.31 in relation to refusal of employment, refusal of employment agency services and detriment under regulations 5, 6 and 9 of the Employment Relations Act 1999 (Blacklists) Regulations 2010;
- 2.32 in relation to the right to request time off for study or training under section 63I of the Employment Rights Act 1996;
- 2.33 in relation to the right to equal treatment, access to collective facilities and amenities, access to employment vacancies and the right not to be subjected to a detriment under regulations 5, 12, 13 and 17(2) of the Agency Workers Regulations 2010;
- 2.34 in relation to the right to a written statement and the right not to be unfairly dismissed or subjected to detriment under regulations 4 and 5 of the Agency Workers (Amendment) Regulations 2019;
 - B-19

- 2.35 in relation to personal injury, of which you are or ought reasonably to be aware at the date of this agreement;
- 2.36 for harassment under the Protection from Harassment Act 1997;
- 2.37 for failure to comply with obligations under the Human Rights Act 1998;
- 2.38 for failure to comply with obligations under the Data Protection Act 1998, the Data Protection Act 2018, the General Data Protection Regulation ((EU) 2016/679) [as it has effect in EU law], or the UK GDPR as defined in section 3(10) and section 205(4) of the Data Protection Act 2018;
- 1.39 arising as a consequence of the United Kingdoms former membership of the European Union; an1.40 in relation to the right not to be subjected to a detriment under regulation 3 of the Exclusivity Terms in Zero Hours Contracts (Redress) Regulations 2015.
- 1.41 arising under any US federal, state or local laws of any jurisdiction that prohibit age, sex, race, national origin, color, disability, religion, veteran, military status, sexual orientation or any other form of discrimination, harassment, hostile work environment or retaliation (including, without limitation, the Age Discrimination in Employment Act of 1967, the Older Workers Benefit Protection Act, the Americans with Disabilities Act of 1990, the Americans with Disabilities Act Amendments Act of 2008, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1866 and/or 1871, 42 U.S.C. Section 1981, the Rehabilitation Act, the Family and Medical Leave Act, the Sarbanes-Oxley Act, the Employee Polygraph Protection Act, the Worker Adjustment and Retraining Notification Act, the Equal Pay Act of 1963, the Lilly Ledbetter Fair Pay Act, the Uniformed Services Employment and Reemployment Rights Act of 1994, the Genetic Information and Nondiscrimination Act of 2008, the Texas Labor Code, Section 1558 of the Patient Protection and Affordable Care Act of 2010, the Consolidated Omnibus Budget Reconciliation Act of 1985, and any other federal, state or local laws of any jurisdiction);
- 1.42 under any other US federal, state, local, municipal or common law whistleblower protection, discrimination, wrongful discharge, anti-harassment or anti-retaliation statute or ordinance;

- 1.44 in respect of any illegal or wrongful acts by any of the Released Parties, including, without limitation, discrimination, retaliation, harassment or any other wrongful act based on sex, age, race, religion, or any other legally protected characteristic; or
- 1.45 regarding any other US statutory or common law matters related to Executives employment or separation from employment with the Company and its Affiliates.

^{1.43} arising under ERISA;

Schedule 3 Advisers certificate

[ON HEADED NOTEPAPER OF ADVISER]

For the attention of [DETAILS]

[DATE]

[[Dear [NAME OF FIRM]] OR [To whom it may concern],]

I am writing in connection with the agreement between my client, [NAME], and [NAME OF COMPANY] (Company) [of todays date**OR** dated [DATE]] (Agreement) to confirm that:

1. I, [NAME] of [FIRM], whose address is [ADDRESS], am [a Solicitor of the Senior Courts of England and Wales who holds a current practising certificate **OR** AMEND AS APPLICABLE].

2. I have given [NAME] legal advice on the terms and effect of the Agreement and, in particular, [its **OR** their] effect on my clients ability to pursue the claims specified in Schedule 2 of the Agreement.

3. I gave the advice to [NAME] as a relevant independent adviser within the meaning of the above acts and regulations referred to at clause 5.4.

4. There is now in force (and was in force at the time I gave the advice referred to above) a policy of insurance or an indemnity provided for members of a profession or professional body covering the risk of claim by my client in respect of loss arising in consequence of the advice I have given them.

Yours faithfully,

[NAME OF ADVISER]

[DATE]

Schedule 4 Announcement

[INSERT THE TEXT OF THE AGREED ANNOUNCEMENT]

Schedule 5 Restrictive covenants [INSERT THE RESTRICTIVE COVENANTS]

Signed by [NAME OF DIRECTOR OR OTHER AUTHORISED SIGNATORY] for and on behalf of [NAME OF COMPANY]

Signed by [NAME OF EXECUTIVE]

[Director OR [POSITION]]



UK Header: EXECUTIVE CHANGES AND STRATEGY UPDATE

VAALCO BOARD MEMBER GEORGE MAXWELL NAMED CHIEF EXECUTIVE OFFICER AND COMPANY UPDATES STRATEGIC FOCUS

HOUSTON – April 12, 2021 – VAALCO Energy, Inc. (NYSE: EGY; LSE: EGY) ("VAALCO" or the "Company") today announced that its Board of Directors has named George Maxwell as Chief Executive Officer effective April 12, 2021. Mr. Maxwell, who has been a Non-Executive Director of the Company since June 2020 and resides in the U.K., will continue to serve as a member of the Board. Cary Bounds who has served as Chief Executive Officer since 2016 will leave VAALCO to pursue other interests. The Company also provided details on its newly updated forward strategy.

The Company's stated goal is to continue to be one of the leading independent exploration and production companies in West Africa, with a strategy of achieving significant shareholder returns by maximizing the value of, and free cash flow from, its existing resources, coupled with highly accretive inorganic growth opportunities. The Board believes that Mr. Maxwell's recent achievement of founding, building, and selling a successful African energy company, as well as his significant experience serving in executive leadership positions, a proven track record in M&A and strong ties to the European, American and African capital market communities will complement and accelerate the Company's ability to achieve this strategy.

The Board believes that enhancing the Company's corporate presence in London, the primary financial and advisory hub for African energy companies, will augment the experience within VAALCO's leadership and complement the Company's supportive equity shareholder base in the U.S. With a majority of transactions for North Africa and Sub-Saharan Africa energy companies taking place within the European financial markets, and a large part of the Company's peer group active within this environment, it is strategically important to have an enhanced presence within this market environment.

Further to VAALCO's successful dual listing on the London Stock Exchange, the Company intends to open a business development office in London and appoint a permanent Chief Financial Officer who will be based in this new London office. Mr. Maxwell expects to split his time equally between Houston and London. The VAALCO executive leadership team intends to maintain its strong relationship with the Gabonese government.

Andrew Fawthrop, VAALCO's Chairman of the Board, commented, "We are excited to begin the next chapter of VAALCO's35-year history with our decision to increase our presence in the U.K. and international markets, a move that we believe will support VAALCO's ambition to maximise shareholder returns. We are very fortunate to have George Maxwell serve as our new CEO as he has significant experience serving in executive leadership positions, a proven track record in corporate and M&A activities and strong ties to the London investment community. He has gotten to know our Company well since joining our Board in June of last year and can therefore ensure a smooth transition of leadership. He has provided invaluable insight and his knowledge of the international energy and capital markets have made him an important resource for us and we look forward to his leadership as our new CEO."

Mr. Fawthrop continued, "We would like to thank Cary for his dedication and valued years of service to VAALCO. Cary guided VAALCO through some very challenging periods for the industry and helped the Company emerge stronger. He oversaw the highly successful 2019/2020 drilling program and the very accretive Sasol acquisition, the combination of which has grown production and reserves considerably and positioned VAALCO for significant cash flow in 2021. We wish Cary the best in the next phase of his career."

George Maxwell commented, "I'm excited by this opportunity to build on a successful foundation and lead the existing teams to grow the Company to the next level. The positions we have in Gabon and Equatorial Guinea provide exciting development opportunities to enhance our business and provide a strong platform for investors.

In this current and future market environment, I believe it is paramount that businesses are sustainable in order to provide benefits to all stakeholders, with a focus on growth and investor returns. This is one of the guiding principles I successfully applied in my previous executive role and I believe VAALCO is in an excellent position to ultimately achieve this. VAALCO has adopted share repurchase programs in the past as a way to return value to our shareholders and we will consider similar programs in the future as well as potential dividends to complement our growth strategy.

I remain wholly committed to our roots and investor base in the U.S., and believe that our expanded presence in the U.K. financial and advisory markets maximizes opportunities for us to achieve our stated strategy."

About George Maxwell

George Maxwell joined VAALCO's Board in June 2020 and has over 25 years of experience in the oil and gas industry, including in both the producing and service/manufacturing arenas. Mr. Maxwell founded Eland Oil & Gas Plc. in 2009 and served as the company's Chief Executive Officer from September 2014 to December 2019, Chief Financial Officer from 2010 to 2014, and as a member of the Board of Directors from 2009 to 2019, until the company was acquired by Seplat Petroleum Development Company Plc for a value in excess of \$500 million, in December 2019. Prior to founding Eland, Mr. Maxwell served as the business development manager for Addax Petroleum and, prior to this, commercial manager in Geneva. Mr. Maxwell joined Addax Petroleum in 2004 and held the general manager position in Nigeria, where he was responsible for finance and fiscal and commercial activities. Prior to this, Mr. Maxwell worked with ABB Oil & Gas as vice president of finance based in the U.K. with responsibilities for Europe and Africa. He held a similar position in Houston, from where the organization ran its operations in ten countries. Mr. Maxwell also serves as a non-executive director on the board of LEKOIL PLC, a publicly traded Africa-focused exploration and production company with interests in Nigeria and Namibia, but plans to step down from this position to focus on his role at VAALCO.

Mr. Maxwell graduated from Robert Gordon University in Aberdeen with a Masters in Business Administration. He is a Fellow of the Energy Institute in the U.K. and has formerly served on the boards of directors of Elcrest Exploration and Production Nigeria Ltd. and Westport Oil Limited.

About VAALCO

VAALCO, founded in 1985, is a Houston, USA based, independent energy company with production, development and exploration assets in the West African region.

The Company is an established operator within the region, holding a 63.6% participating interest in the Etame Marin block, located offshore Gabon, which to date has produced over 120 million barrels of crude oil and of which the Company is the operator.

For Further Information

VAALCO Energy, Inc. (General and Investor Enquiries)	+00 1 713 623 0801
Website:	www.vaalco.com
Al Petrie Advisors (US Investor Relations)	+00 1 713 543 3422
Al Petrie / Chris Delange	
Buchanan (U.K. Financial PR)	+44 (0) 207 466 5000
Ben Romney / Kelsey Traynor / James Husband	VAALCO@buchanan.uk.com

Forward Looking Statements

This document includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. All statements, other than statements of historical facts, included in this document that address activities, events, plans, expectations, objectives or developments that VAALCO expects, believes or anticipates will or may occur in the future are forward-looking statements. These statements may include statements related to the impact of the COVID-19 pandemic, including the sharp decline in the global demand for and resulting global oversupply of crude oil and the resulting steep decline in oil prices, production quotas imposed by Gabon, disruptions in global supply chains, quarantines of

our workforce or workforce reductions and other matters related to the pandemic, well results, wells anticipated to be drilled and placed on production, future levels of drilling and operational activity and associated expectations, the implementation of the Company's business plans and strategy, prospect evaluations, prospective resources and reserve growth, VAALCO's 2021-2022 drilling program, its activities in Equatorial Guinea, expected sources of and potential difficulties in obtaining future capital funding and future liquidity, its ability to restore production in non-producing wells, future operating losses, future changes in crude oil and natural gas prices, future strategic alternatives, future acquisitions, capital expenditures, future drilling plans, interpretation of seismic data and costs thereof, negotiations with governments and third parties, timing of the settlement of Gabon income taxes, and expectations regarding processing facilities, production, sales and financial projections. These statements are based on assumptions made by VAALCO based on its experience and perception of historical trends, current conditions, expected future developments and other factors it believes are appropriate in the circumstances. Such statements are subject to a number of assumptions, risks and uncertainties, many of which are beyond VAALCO's control. These risks include, but are not limited to, crude oil and natural gas price volatility, the impact of production quotas imposed by Gabon in response to production cuts agreed to as a member of OPEC, inflation, general economic conditions, the outbreak of COVID-19, the Company's success in discovering, developing and producing reserves, production and sales differences due to timing of liftings, decisions by future lenders, the risks associated with liquidity, lack of availability of goods, services and capital, environmental risks, drilling risks, foreign regulatory and operational risks, and regulatory changes.

Investors are cautioned that forward-looking statements are not guarantees of future performance and that actual results or developments may differ materially from those projected in the forward-looking statements. VAALCO disclaims any intention or obligation to update or revise any forward-looking statements, whether as a result of new information, future events, or otherwise.